Exploring the Role of Long Duration Electricity Storage in Ensuring Adequacy

amid High Renewable Penetration and Future Climate Uncertainties Federica Benisi Master Thesis

Master Thesis September 30, 2024

Contents

Li	st of I	igures	V
Li	st of [·]	ables	v
At	ostrac	t	i
Pr	eface	i	i
1	Intro	duction	1
2	Lite		3
	2.1	2.1.1 General Introduction	-
	2.2	2.1.4 Current challenges () Meteorology and Resource Adequacy Failures () 2.2.1 General Introduction () 2.2.2 Research trends () 2.2.3 Current Challenges ()	7 7
	2.3 2.4 2.5	Energy Storage for Net-Zero and Resource Adequacy. 1 2.3.1 General Introduction 1 2.3.2 The Design Landscape of Long-Duration Electricity Storage 1 2.3.3 Technological Options for Long-Duration Electricity Storage 1 Current Challenges 1 Knowledge Gap and Research Question 1	1 3 4 5
3	Metl 3.1 3.2	ods of Approach18Energy System Modelling183.1.1Calliope183.1.2Modelling Long Duration Energy Storage203.1.3Time Clustering20Monte Carlo Methods213.2.1State Duration Sampling Approach21	8 8 0 0
4	Sys 4.1 4.2 4.3	em Modelling and Analysis23Methodology Overview.24Model Outline.244.2.1 Model Assumptions244.2.2 Input Data.244.2.3 Geographic Scope, Transmission Grid and System Scale264.2.4 Time Resolution and Horizon264.2.5 Generation274.2.6 Short-Term and Long-Term Storage364.2.7 Electrical Load37Adequacy Assessment.384.3.1 Selection of Climate Year364.3.3 Storage Model Configurations374.3.4 Calculation of EENS36	3445667015567

5	Results5.1Generation and Storage Capacity Expansion.5.2Analysis of LDES Dispatch Modeling5.3Impact of LDES Discharge Duration on System's Adequacy.5.4Varying power capacity, comparison with SDES	43 46
6	Discussion of the Results6.1Project background and scope.6.2Model and Simulations Findings.6.3Socio-Economic Impact6.4Limitations	54 56
7	Conclusion and Recommendations 7.1 Conclusion 7.2 Answer to Research Question	
A	Appendix A.1 Impact of LDES Discharge Duration on RA - Load Sheddig Distribution	62 62
Bil	bliography	64

List of Figures

2.1 2.2	Energy Storage Technologies	13 13
3.1	Calliope Conceptual Scheme	19
4.1 4.2	Model Framework	24
4.2 4.3	Transmission Grid of the model	26 28
	Future and past temperature in the Netherlands	
4.4	Future and past solar capacity factor in the Netherlands	29
4.5	Future and past wind capacity factor in the Netherlands	29
4.6	Future and past demand in the Netherlands	32
4.7	Load curtailment implementation	34
4.8	Extreme meteorological conditions of 1985. Source: [1]	36
4.9	Dunkelflaute example	36
4.10	Operating history of a generating unit	37
5.1	Installed generation capacities for Europe in 2050	40
5.2	Technology-specific storage power for Europe in 2050.	41
5.3	AC transmission in 2050	41
5.4	Seasonal peak net load frequency	42
5.5	Yearly peak net load frequency	43
5.6	Hydrogen Dispatch - Case 1 - E2P = 100	44
5.7	Comparison of LDES dispatch with varying look-ahead periods	45
	Annual national dispatch results in 2050 in NLD - 1986 case	46
5.9	Net Electricity Demand in 2050 in NLD - 1986 case	46
	Loss of load in Germany	47
	Wind power output - NLD	48
	Dispatch details and hydrogen cycles for an E2P of 50 hours.	49
	Dispatch details and hydrogen cycles for an E2P of 100 hours.	49
	Dispatch details and hydrogen cycles for an E2P of 50 hours.	50
	Load Shedding Distribution - Case of halved LDES power capacity	51
	Load Shedding Distribution - Case of LDES energy capacity reduced to 1/4th.	52
	Load Shedding Distribution - Case of halved LDES energy capacity.	52
6.1	Economic equilibrium determining the reliability standard	56
A.1	Load Shedding Distribution - E2p = 50 hours	62
	Load Shedding Distribution - E2p = 100 hours	63
	Load Shedding Distribution - E2p = 200 hours	63

List of Tables

	Comparison between resource adequacy metrics	
	storage	15
	Climate models used by [2]. Data retrieved from [3]	
	EENS and NFC values for various LDES dispatch strategies.	
6.1	Frequency of load shedding duration for all cases	56



amid High Renewable Penetration and Future Climate Uncertainties

by



to obtain the degree of Master of Science in Sustainable Energy Technology at the Delft University of Technology.

Authors: Project duration: Supervisors: F. Benisi 5755107 December 4, 2023 – September 26, 2024

Defense: Thesis committee: Dr. S. H. Tindemans Dr. ir. K. Bruninx ir. J. W. Peterse September 26th, 2024 TU Delft, Chair, Supervisor TU Delft Guidehouse, Supervisor

An electronic version of this thesis is available at http://repository.tudelft.nl/.





Abstract

In the context of increasing renewable energy integration and the push for decarbonization, this research addresses the need for long-duration electricity storage to ensure the reliability and adequacy of the future European power system. The variability of renewable energy sources such as wind and solar, coupled with the challenges posed by climate change, requires advanced storage solutions to balance periods of low energy generation and high demand.

This thesis explores how long-duration energy storage (LDES) and short-duration energy storage (SDES) can effectively contribute to resource adequacy by assessing various operational strategies, storage capacities, and system conditions. Using the Calliope energy system modeling framework, the research simulates a future interconnected European grid in 2050, considering different weather conditions and storage configurations to evaluate metrics such as Expected Energy Not Served and Loss of Load Expectation.

The results demonstrate that increasing the energy storage capacity and discharge duration of LDES significantly reduces LOLE, particularly during extended periods of low renewable output. However, SDES plays a complementary role, addressing short-term reliability challenges by reducing EENS during shorter disruption events. This combination highlights the need for a balanced deployment of both SDES and LDES to optimize grid performance and ensure resource adequacy. Through these findings, the thesis contributes valuable insights for energy planners and policymakers, aiming to foster a resilient, low-carbon electricity grid capable of meeting the demands of a renewable-dominated future.

Preface

I would like to use this space to thank all the people who collaborated in achieving my goals and in my academic and personal journey.

I would like to thank TU Delft and my thesis committee members. I am thankful to Dr. Kenneth Bruninx for sharing his perspective and expertise with great availability, which helped me improve this project. My deepest thanks go to my main supervisor, Dr. Simon Tindemans for guiding me in this process and helping me overcome the main difficulties of this project. I am also deeply grateful to my daily supervisor and company's buddy, Jaap Peterse and Eugenia Morino, who provided constant support and friendship. Finally, I would like to thank the entire Guidehouse Utrecht office for making their resources available to me and offering me the opportunity to get to know the world of consultancy more closely.

I would like to thank my friends, both those I met in Italy and who are now scattered around the world. You have stood by me through many stages of my life, and despite the years and distance, you are still there for me. In particular, I would like to thank Andrea and Elena. We have experienced many different phases in our friendship, and it is almost strange to now find ourselves talking together about home, work, and things that seemed so far away from us when we first met. Yet, the wisdom you give me every time has never changed. I would like to thank all the friends I met in the Netherlands. Moving here to continue my academic journey has been one of the best decisions of my life, and a large part of that is because of you. Each of you has contributed something special to my experience, whether it was laughter, a serious conversation, or a lesson about myself. In particular, I want to thank Elena, Margherita, and Ilaria for being there for me during good or difficult times. Your sensitivity and understanding have made me feel like I have found a family abroad. I would also like to thank Gregory for bringing joy into my life nearly every day over the past year, with your presence I have never missed the sun in this new country.

I would like to thank my family, especially my sister Beatrice and my mother, for supporting me and putting up with me throughout these years of study. The most special thanks to my Nonno Pino and Nonna Gioia, you have left the biggest mark on me with your kindness, curiosity, and passion for life and people. You are not here today, but I think of you every day, and your memory has given me strength and inspiration throughout this journey.

Federica Benisi Delft, September 2024

Nomenclature

BPS	Bulk power system
CAES	Compressed air energy storage
CCS	Carbon capture and storage
COP282	United Nations Climate Change Conference
DE	TYNDP Distributed Energy scenario
DESSTinEE	Demand for Energy Services, Supply and Transmission in EuropE
DF	Dunkelflaute
E2P	
	Energy-to-Power
ECMWF	European Centre for Medium-Range Weather Forecasts
EENS	Expected Energy Not Served
EERA	European Resource Adequacy Assessment
ENTSO-E	European Network of Transmission System Operators for Electricity
FO	Forced Outage
FOR	Forced Outage Ratio
GA	TYNDP Global Ambition scenario
GCM	General Circulation Models
HDD	Heating Degree Days
LCOE	Levelized cost of electricity
LDES	Long Duration Electricity Storage
LOLE	Loss Of Load Expectation
LOLP	Loss Of Load Probability
MC	Monte Carlo
MSLP	
	Mean sea level pressure
MT	Medium-term
MTTF	Mean-time-to-failure
MTTR	Mean-time-to-repair
NFC	Number of Full Cycles
NT	TYNDP National Trend scenario
P2G	Power-to-Gas
P95	95th Percentile
PGP	Power-to-Gas-to-Power
PHES	Pumped hydro energy storage
RA	Resource Adequacy
RES	Renewable energy source
SDES	Short-duration Energy Storage
SOC	State-of-charge
TES	Thermal energy storage
TSO	Transmission System Operator
TTF	Time-to-failure
TTR	Time-to-repair
TYNDP	Ten-Year Network Development Plan
VOLL	Value Of Lost Load
VRES	Variable renewable energy sources
AUT	Austria
BEL	Belgium
DEU	Germany
DNK	Denmark
FRA	France
GBR	Great Britain
NLD	The Netherlands

Introduction

To keep global warming to no more than 1.5 °C as called in the Paris Agreement, the transition to net-zero emissions must be fully complete by 2050. On this matter, in December 2023 the United Nations Climate Change Conference (COP28) closed with an agreement that signals the "beginning of the end" of the fossil fuel era by laying the ground for a transition underpinned by deep emissions cuts and scaled-up finance [5]. Variable renewable energy sources(VRES) such as solar, wind, and hydropower will lead this transition and gain an increasing share in the power system, thanks to their economic competitiveness compared to fossil fuels. If on one side this represents a big achievement in the way of decarbonization, on the other side the intermittent nature of these sources due to climate dependency poses considerable challenges for energy planners and grid operators. Moreover, the electrification of different energy demand sectors (heating, industry, transport) will also result in an increased dependency of the electricity demand on weather conditions. For example, ENTSOe has identified for the winter of 2023 a small regional adequacy risk in and around France, with the highest probability of affecting Belgium and Great Britain in the event of an extreme cold spell, driven by the sensitivity of the demand in France to temperature and are present only under extreme weather conditions combined with high unplanned outages. [6].

Achieving reliable and affordable electricity systems may revolve around the capability of storing substantial amounts of low-cost energy over long durations. As renewable energy sources and shortduration electricity storage technologies (SDES) become more prevalent in the market, and as efforts towards deeper decarbonization intensify, long-duration electricity storage technologies (LDES) are gaining appeal to manage prolonged mismatches between energy supply and demand. These technologies can economically store energy for durations ranging from intra- to inter-day periods, bridging the gap between current short-duration batteries and seasonal storage The ability to provide energy during prolonged periods of low or no renewable generation allows LDES to support the three key goals of the energy transition: enhancing security, improving affordability, and reducing emissions from power supply [7].

LDES also has the potential to contribute to the resource adequacy of a power system, which is the ability to meet the energy demand at all times [8]. These technologies help balance the grid during peak demand periods, ensuring that the system has sufficient resources available. As power systems become more reliant on VRES, utilities will increasingly need to assess the role of LDES in maintaining resource adequacy and capacity value, the contribution a given generator makes to overall system reliability and adequacy [9].

Despite the promising potential of LDES, several challenges slow down its deployment. While pumped hydro energy storage (PHES) and compressed air energy storage (CAES) are commercially available at scale, advanced technologies like electrochemical storage are still in development, with limited demonstration at utility scale [10]. High initial costs, technological uncertainties, regulatory barriers, and the lack of established market mechanisms are among the primary obstacles that need to be addressed. For LDES to realize its full potential, it will require long-term vision and sustained investment

from both public and private sectors, ensuring that it becomes a key element in a future electricity grid. Understanding the capacity value drivers of LDES is crucial for developing policies and strategies that can accelerate its deployment and integration into the energy system.

The above leads to the following research question:

What are the capacity value drivers of long-duration electricity storage for an interconnected European power system in 2050 in a context of high penetration of renewables and future climate?

The research question is broken down using the following sub-questions:

- How is EENS affected by the operational profile of LDES?
- How is EENS affected by the discharge duration of LDES?
- · How is EENS affected by LDES installed capacity and how does LDES interact with SDES?

Report Outline

The report is structured as follows:

Chapter 2 presents the current state-of-the-art knowledge relevant to the topics addressed in this research, including resource adequacy assessments, the growing dependence of power grids on VRES and consequently the influence of meteorological conditions on resource adequacy shortfalls, and the need for flexibility solutions.

Chapter 3 introduces the main methods used by energy system modelers to reproduce the power system's random behavior and conduct system adequacy assessments through Monte Carlo simulations.

Chapter 4 explains how the methods previously introduced are used to build a European interconnected power system of the future and assess the contribution of LDES to the system's resource adequacy.

Chapter 5 analyses the simulation results.

Chapter 6 outlines and discusses the findings of this research and presents its limitations.

Chapter 7 presents the research conclusions, addressing the main research question.

 \sum

Literature Review

This literature review aims to present the current state-of-the-art knowledge relevant to the topics addressed in this research. Section 2.1 explores frameworks used in resource adequacy assessments. Section 2.2 examines the impact of meteorological conditions on resource adequacy failures. Section 2.3 discusses various flexibility options that enhance grid reliability, with a particular emphasis on longduration electricity storage solutions. Following these discussions, Section 2.5 identifies the existing knowledge gaps and outlines the research questions that this thesis aims to address.

2.1. Resource Adequacy Framework

One of the European Union's main objectives in energy policy is to provide secure, affordable and clean energy [11]. In agreement with this objective, energy planners and grid operators work to guarantee system security and adequacy. The concepts of system security and system adequacy are explained in Section 2.1.1, together with the main figures used to assess the adequacy of a system presented in Section 2.1.2. Finally, Section 2.1.3 presents a comparison between these metrics, while Section 2.1.4 shows the main downsides of traditional resource adequacy metrics.

2.1.1. General Introduction

"Security of Electricity Supply" stands as one of the fundamental aspects of the European Union's strategy for climate and energy [12]. Various aspects are considered in this matter, including system adequacy. Adequacy is defined as "a measure of the ability of a bulk power system to supply the aggregate electric power and energy requirements of the customers within component ratings and voltage limits, taking into account scheduled and unscheduled outages of system components and the operating constraints imposed by operations" [8]. While security deals with the operational aspects, adequacy is integral to the system's development and planning [12]. It involves the ability to meet long-term demand while considering supply-demand uncertainties and the extended time required for the capacity of network expansion.

Some of these uncertainties are gaining significance, such as correlated generator outages and weather variations [12]. Grid planners address these uncertainties by statistically evaluating them to project power grids' resource needs to reach an acceptably low level or risk of capacity shortages. Making an adequacy evaluation of the system can encompass these aspects [12]:

- Generation adequacy: assessing the capacity of the generation within a power system to meet its consumption requirements
- Transmission adequacy: evaluating the power system's ability to handle the power flow from generation to consumption centers.
- System adequacy: takes into account both generation and transmission adequacy.

Risk metrics can then be used to determine the necessary investments in power grids and the amount and type of generation to be constructed. The optimal adequacy should balance the investments needed to guarantee supply on every occasion and the cost of the energy not supplied to the customers [12].

European adequacy evaluations are conducted by the European Network of Transmission System Operators for Electricity (ENTSOe), the association for the cooperation of the 40 European Transmission System Operators (TSOs) who represent 35 countries and are responsible for the secure and coordinated operation of Europe's electricity system. This study aims to assess the generation adequacy of the European interconnected system, with a simplified approach to transmission modeling. Typically,

2.1.2. Adequacy Metrics

In evaluating generation adequacy, the underlying idea is that achieving absolute adequacy in an electrical system is practically unattainable without considering exorbitant investments. For this reason, in reality, the aim is to strike a balance between cost-effectiveness and reliability: system failure occurrences are tolerated when the resulting inconveniences for customers remain manageable or when there's a reluctance to invest more in enhancing reliability. Determining the acceptable level of unserved energy is typically done through reliability criteria. These metrics benchmark a system's readiness compared to a reference value called adequacy standards. By using adequacy indicators and comparing them to a reliability standard, it becomes possible to compare different systems and determine if a power system's level of adequacy is acceptable.

The European Commission describes the main adequacy indicators utilized in the European electricity market in [12]. This report aims to introduce the concept of adequacy and the main methodologies used in industry to assess the adequacy of a power system in Europe. The following sections present the main indicators of system adequacy with definitions found in this report.

Expected Energy Not Served (EENS) [GWh/year]

EENS represents the amount of electricity demand that is not met by the available generation plus imports [12]. Being a probabilistic metric [13], it can be obtained as an average of the Energy not Served (ENS) found across the simulations that are conducted [14], as described by the formula given by ENTSO-e in [15]:

$$EENS = \frac{1}{N} \sum_{j \in S} ENS_j$$
(2.1)

Where N is the number of simulations and ENS_i is the ENS of the j^{th} simulation [15].

Load Of Load Expectation (LOLE) [h/year]

LOLE measures the number of event periods, where an event period is a specified period (day, month, year) in which one or more shortfall events occur [16]. ENTSO-e defines LOLE as the average number of hours per year in which the available generation plus import cannot cover the load in an area or region [14]:

$$LOLE = \frac{1}{N} \sum_{j \in S} LLD_j$$
(2.2)

where LLD_j is the loss of load duration associated with the j^{th} simulation and N is the number of simulations [15].

Loss Of Load Probability (LOLP)

The Loss Of Load Probability over a certain period represents the likelihood of experiencing at least one shortfall during that period, essentially indicating the probability of that being an event period. Although LOLE and LOLP are conceptually separate metrics, if the same period definition is applied to both, it is feasible to compute one metric in terms of the other.

95th Percentile (P95) [h/year]

The P95 values represent the 95th percentile, indicating the LOLE value that is near the maximum observed across all simulations, including the effects of weather and outages on the power supply. [17].

2.1.3. Comparison between metrics

Quantifying the EENS is the most straightforward method for monetizing interruption costs to compare possible investments aimed at achieving adequacy targets. In fact, according to [12] to assess the economic value of supply reliability or the financial impact resulting from interruptions, relevant cost metrics should be expressed as \in /interruption, \in /kW of peak load, \in /kWh of annual energy consumption, or \in /kWh of ENS.

While LOLE is commonly used due to its simplicity, it lacks information about the severity of the issue, as it is possible to get different LOLE while keeping the EENS constant [18]. For instance, a system-wide blackout or small demand reductions caused by the inability to meet peak demand can result in the same total number of hours of LOLE [12]. This can lead current metrics to overlook the considerably greater impact of these large events on consumers and society as a whole.

In [12], the European Commission outlines the key characteristics of each metric discussed, providing readers with guidance on when it may be appropriate to use each one.

Metric	Advantages	Disadvantages
EENS	 Quantification of ENS: possibility of monetizing interruption costs. It accounts for the magnitude of shortages. 	 A comparison between electricity systems of different dimensions is not possible but with a normalization of EENS.
LOLE	 Simple definition of the probability of encountering problems. Widely used and accepted in the industry, facilitating benchmarking. Focus on the frequency of outages. 	 The magnitude of the issue is not measured, as it depends on how the electricity system is operated under critical conditions. Lack of economic perspective.
LOLP	 Simple way to define the like- lihood of coming across prob- lems. Option to compare electricity systems of varying sizes. 	 The magnitude and duration of the issue are unquantified. Lack of economic perspective.
P95	 It shows extreme values, helping in understanding worst-case scenarios Useful for planning against high-impact, low-probability events 	 If there are very extreme outliers in the simulation data, the P95 value can be significantly influenced by these ones, potentially skewing the perception of risk. Focuses on extreme cases and might not represent typical performance, so it may lead to a disproportionate focus on high-risk scenarios.

Table 2.1: Comparison between resource adequacy metrics. Retrieved from [12]

2.1.4. Current challenges

Traditional resource adequacy metrics have been criticized in several ways [19]. First, the existing metrics focus on a single aspect of energy shortfalls, such as their frequency, duration, or magnitude. However, there is a growing recognition of the need to transition towards compound metrics or multiple metrics that offer a more comprehensive understanding of the adequacy challenges faced by the system. These broader metrics could better inform the types of resources required to address specific types of energy shortfalls. Secondly, while current metrics provide an average or aggregate characterization of shortfalls, they lack information on the distribution of frequencies, durations, or magnitudes of individual events. While using multiple metrics could address some limitations, there's still valuable insight into the distribution of shortfall events that are commonly not captured.

2.2. Meteorology and Resource Adequacy Failures

The upcoming energy transition will enhance the role of renewable energy sources in future power systems. Using intermittent sources will also increase the dependency of power grids on the weather. It is therefore important to consider the impact of meteorology on resource adequacy. While the main current situation and research trends in this matter are presented in respectively Section 2.2.1 and Section 2.2.2, Section 2.2.3 presents current challenges when taking into account climate variability in resource adequacy assessments.

2.2.1. General Introduction

Policies promoting low-carbon energy generation have significantly increased renewable energy production, displacing traditional thermal sources. According to [12], owing to their low operating costs, renewables now rank higher than flexible thermal plants in the merit order. Consequently, there has been a substantial decrease in the utilization of thermal plants, with their average utilization rate dropping from 50% to 37% between 2008 and 2013 [20]. This decline in flexible thermal generation, crucial for managing fluctuations of renewable energy production, poses challenges in ensuring grid adequacy in electricity markets.

While in traditional resource adequacy (RA) assessments resources are predominantly dispatchable, in the future resources will become predominantly non-dispatchable [19]. As a consequence, instead of focusing on probabilities of discrete independent mechanical or electrical failures, weather-influenced correlated events should now be recognized as a driving factor of reliability. In traditional resource adequacy analysis, periods with a higher likelihood of shortages were typically associated with peak loads, as variable resources constituted a small portion of the resource mix. However, the periods posing a risk of shortage are now shifting. For example, in the case of solar power, the diurnal pattern causes a decline in solar production at the end of the day. Additionally, extended cloud cover can further reduce solar output as storms pass through a region. Similarly, for wind generators, wind speeds can be correlated as different atmospheric conditions or storm fronts move through a region.

There is a growing discussion on the need for new flexible resources to support the variability of renewable generation [12]. Renewable energy sources can experience rapid and unpredictable changes. To meet demand at all times, these supply variations can be addressed through quick-responding generation, energy storage, or demand adjustments [12]. These elements underline the importance of considering weather when considering the resource adequacy of the power grid and the potential need for a source of flexibility.

2.2.2. Research trends

Reliability partly depends on maintaining resource adequacy, which pertains to the system's ability to consistently match electricity supply (or generation) and demand despite unforeseen circumstances. In this matter, meteorological conditions can contribute to disruptions by straining generation capacities, impairing transmission networks, and amplifying fluctuations in demand. RA failures, i.e., times where demand exceeds supply operationally at bulk power systems (BPS) level, are often responsible for large-scale rolling outages, e.g., in California in 2020 [21] and Texas in 2021 [22]. These two events were caused by a combination of higher than anticipated demand, due to respectively a heatwave and a cold snap, and generator outages driven by extreme weather. In Europe in 2003, due to a hot and dry summer, demand seriously threatened to exceed supply in several countries [23]. More recently, in 2022, hydro generation hit a record low due to extreme droughts, and nuclear generation dropped to the lowest output in 30 years due to unexpected outages at French nuclear plants [24].

All of these events required action, such as implementing rolling outages, by the system operator to avoid severe consequences for the system. Therefore, recognizing the primary meteorological factors contributing to power system strain is crucial for policymakers, grid operators, and energy stakeholders to enhance the future European power grid in the face of a growing reliance on VRES and climate unpredictability.

From the literature, three main fields of research are found at the intersection between meteorology

and resource adequacy:

- The main meteorological drivers of European power system stress
- · The increasing penetration of wind and solar power
- The non-static meteorology caused by natural variability and anthropogenic climate change

These trends and their research gap and challenges are investigated in the sections below. This investigation is particularly relevant to the research presented in this report because it directly addresses the critical issues of including meteorology in resource adequacy assessments.

Main meteorological drivers of European power system stress

Due to the variability of renewable, weather-dependent energy sources across different temporal scales, it is crucial to understand how meteorological conditions influence this variability and, consequently, power system stress. This understanding is essential for the efficient operation of a power system. Peak demand is the highest amount of electric demand in a particular period. In contrast, net demand is the total electricity demand in the system minus utility-scale wind and solar generation. For this reason, daily peak net demand can occur at a different time than peak demand [25]. Most of the literature studies the meteorological conditions and weather patterns that lead to peak and net demand by correlating particular weather variables of a region to the electricity demand. Temperature is the variable that is mostly correlated to part of the demand [26], for example for heating and cooling. According to [27], since weather data typically have a longer and more consistent historical record than electricity demand data, working directly with weather data would be more advantageous to analyze its relationship with electricity demand instead of relying solely on observed demand data. This approach allows for a deeper exploration of how weather conditions influence demand patterns and system stress.

In countries with a winter demand peak, the availability of wind can be uncertain due to different meteorological conditions. Several studies have examined the potential for the availability of wind generation at peak demand, with the term "low wind, cold snap" being used in the literature to describe times of potential concern for winter peaking energy systems [28]. Another extreme meteorological event is the so-called anticyclonic gloom described by the German term "Dunkelflaute" (DF). Generally, this event is characterized by the simultaneous occurrence of calm winds and dense cloud cover in winter. According to [29], an event is classified as DF if the solar and wind capacity factors are below 20% of the historical average for more than one day. Bloomfield [30] examines mean sea level pressure (MSLP), 100 m wind speed, 2 m temperature, and surface solar radiation anomaly composites during Europe's ten highest peak load events to identify the typical meteorological conditions associated with such events. The study reveals that during peak demand periods, Europe tends to experience high-pressure systems, leading to abnormally cold temperatures in the region. High-pressure systems are areas in the atmosphere where air is sinking, leading to stable and dry weather conditions. Large negative near-surface wind speed anomalies were found in northern France, Belgium, the Netherlands, northern Germany, and the UK, and even larger negative anomalies offshore in the North Sea, resulting in below-average wind power generation throughout the UK, Netherlands, Belgium, and Germany.

In his analysis regarding GB's weather, Dent [27] agrees with the result, confirming that the presence of this high-pressure system is observed consistently during severe weather events or periods of extreme conditions. High-pressure weather conditions usually present calm, clear, sunny conditions and consequently an above-average surface solar radiation [30], so one could think that the low wind power output could be compensated by solar power. However, Pfenninger [31], which presents the correlation between national wind and PV output for simulated future 2030 fleets in the UK, shows that the increased capacity of wind energy by 2030 contributes to heightened variability, imposing strain on the rest of the power system. Conversely, PV generation tends to lag during winter months, exacerbating the challenge of meeting increased energy demand during this time.

Another event associated with times of power system stress is the so-called "heat wave". In summer, maximum electricity demand for cooling is needed, but from the resources point of view, summer wind presents a minimum on its annual cycle, so a combination of maximum electricity demand can coincide with a minimum of wind power production. Molina et al. [32] shows that during heat wave events, an increase of 3.5%-10.6% in electricity demand and a decrease up to -30.8% in wind power production is obtained, with variability depending on the country. These results therefore add to the need for new flexibility options such as interconnection, storage, and load-shifting to help balance and harness this output.

The increasing penetration of intermittent generation

Wind and solar power generation are inherently variable and dependent on meteorological conditions such as wind speed, solar radiation, and cloud cover. Unlike conventional fossil fuel-based generation, which can be dispatched as needed, renewable energy generation is subject to weather patterns. Wind speeds and solar irradiance exhibit significant spatio-temporal variability and forecast errors [33]. As a result, the availability and output of wind and solar power can fluctuate widely over short time scales, leading to increased uncertainty in electricity supply and demand balance and a more complicated framework to do RA assessments. Since wind and solar power are a function of wind speeds and solar irradiance, increasing wind and solar power penetrations will increasingly link electricity supply and demand to these meteorological variables. For this reason, it becomes imperative to understand their variability and impact on the power system.

To keep pace with the increasing penetration of renewable power generation, energy modelers have been moving from using average annual or seasonal capacity factors to time series data [31]. In this way, it is possible to capture periods of power system stress, for example when both wind and solar power output are low. The first studies on this matter tended to use a single or typical year of generation data, however overlooking the substantial year-to-year fluctuation in weather or considering a single country in isolation, which ignores the possibility of balancing renewable intermittency through international connection [34]. Collins et al. [35] use a continental electricity system planning model and 30 years of hourly wind and solar data to determine the impact of long-term weather patterns on the operation of the European electricity system. The results show that the variability of CO2 emissions and total generation costs for this system could increase 5-fold by using multiple years. Moreover, the study finds a strong correlation between VRES penetration and increased variability in the operation of conventional generators, primarily those that offer baseload power, playing a key role in balancing resource availability. However, storage technologies such as pumped hydro and battery storage capacity are not included in the study and the influence of weather on power demand is not considered.

For a case study in Britain, Staffell et al. [36] combines an analysis of weather-affected wind and solar generation and demand over multiple decades in an attempt to capture a fuller range of weather events. This helps to quantify the variability of the net demand that will have to be supplied by conventional generators. The study finds that the year-to-year variability of net electricity demand will increase by 80% by 2030 and that selecting a single year rather than 25 historical can result in a minimum net demand ±13% away from the 'true' mean. This underscores the importance of considering multiple years to achieve more reliable and representative results in analyzing electricity demand trends. On the other hand, Dent [27] suggests that historical weather records may be influenced by a few specific years, potentially skewing our understanding of past weather patterns. Longer records tend to reveal more extreme outlier events, such as harsh winters in the 1980s or specific years like 1962-1963 and 1946-1947. In summary, the literature underscores the need for careful interpretation and consideration of historical data in understanding past weather patterns and trends.

Non-static meteorology caused by natural variability and anthropogenic climate change

The system transformation of the energy sector needed to reach the European sustainability goals will increasingly couple energy supply and demand with weather and climate. Energy system modelers have succeded in representing the impact of present-day weather and climate to guarantee a reliable energy system that is resilient, affordable, and sustainable [37]. This means taking into account the variability and uncertainty of the present weather and climate. However, there is still a disconnect between the energy modeling community and climate modelers regarding what concerns the representation of the future climate [37].

Several studies in the literature have pointed out the different concerns that non-stationary meteorology

imposed by anthropogenic climate change and long-term variability may impose on the energy system. A change in the meteorological and hydrological conditions could affect the whole energy chain. From the supply side, renewables will likely be affected by climate change. In particular hydropower and thermoelectric generation in Europe could decrease by up to 20% in a +3 °C global warming scenario according to [38]. Moreover, the temporal characteristics of wind power generation can be subject to climate change, leading to a higher probability for long periods of low wind power generation and a stronger seasonal wind variability [39] [40]. This greater variability could cause a higher frequency of energy droughts, mainly a sequence of days with energy production below a threshold or with a significant mismatch between VRES production and demand [41]. Moreover, extreme temperatures and weather events may cause big fluctuations in the hourly peak electricity demand of the system needed for heating and cooling [42] or infrastructural damage to the power grid [43].

As extreme weather driving these events (e.g., heat and drought) is projected to increase in severity and/or frequency under climate change [44], taking into account climate change becomes necessary not only for climate modelers but also for energy planners that want to guarantee the reliability and resilience of an energy system. European Resource Adequacy Assessment EERA 2023 takes into account climate change scenarios [45] in a simplified manner using a transitionary solution while an enhanced and forward-looking Pan-European Climate Database for future resource assessments is currently under development.

2.2.3. Current Challenges

As the energy transition entails a higher dependence of the electricity sector on meteorology, the availability and the use of high-quality meteorological data in energy system modeling becomes more and more important. Weather variables such as temperature, wind speed and solar irradiation are important to model the demand and renewable generation of a system. It is well documented the correlation between demand and temperature, with high temperatures resulting in increased demand for cooling and lower temperatures in increased demand for heating [26]. Solar and wind power generation can be modeled using solar radiation and wind speed data. Wind power relates to wind speed at wind turbine hub height and solar power generation predominantly relates to the amount of incoming solar radiation on the solar panel. However, climate model data are usually difficult to interpret and non-accessible to energy system planners, for example, due to the need for interpretation and calibration of a large volume of weather data. Bloomfield et al. [2] outlines the requirements of a dataset that can be properly used by planners in their analysis:

- They need hourly resolution: Power system planning models need to be run at a high temporal resolution such as hourly, to consider potential operational constraints and change in generation and demand patterns.
- · They use a multi-decadal historical period to account for the impact of inter-annual variability
- · They take climate change into account

Several weather variable datasets are available in the literature that satisfy the first two requirements, such as Copernicus Climate Data Store [46] and Renewables Ninja [47]. However, most of the datasets are based on historical observation and do not take into account future climate projections. Alternatively, datasets can be obtained through General Circulation Models (GCMs) which represent physical processes in the atmosphere, ocean, cryosphere and land surface. These models are the most advanced tools currently available for simulating the response of the global climate system to increasing greenhouse gas concentrations [48]. However, the use of climate models for present and future climate presents its challenges. First, while they include the impact of climate change on the weather, they do not have the temporal resolution required, as they are usually presented with a 3-hourly or daily resolution [38] [39] [40]. Second, there is the fact that climate models have their limitations and uncertainty. GCMs and climate models are probabilistic and can show a range of projections that could change depending on the particular model that is used. For example, for many regions projected change in rainfall varies from an increase in some GCMs to a decrease in others [49]. On this matter, Bloomfield et al. [50] underscores the importance of considering multiple climate models and projections, especially when addressing extreme generation scenarios like wind/solar droughts. The diverse responses in future climate projections emphasize the critical need for power system planners to avoid relying solely on multi-model mean climate projections (i.e., the average of multiple model simulations) and instead utilize these findings to establish a confidence interval of likely outcomes. Third, GCMs do not reproduce accurately some important drivers of weather in Europe related to the boundary of air masses that play a significant role in shaping regional weather patterns. In particular, they do not reproduce well storm tracks across Europe, at a latitude of 50-60° and areas of high pressure, which are important drivers of the power system stress as described in Section 2.2.2 [27].

As climate change does not imply the recurrence of cold spells again, there can be gaps in understanding the full spectrum of risks and potential outcomes associated with climate change. For these reasons, historical climate data and scenarios can provide valuable insights into potential future climate conditions and provide valuable and understandable insights to the community of electricity system planners and policymakers. These scenarios may need to be reweighted or adjusted to account for the likelihood of climate change impacts, similar to how the severe winter of 1962-1963 is considered in assessing future risks [27].

2.3. Energy Storage for Net-Zero and Resource Adequacy

The previous sections introduced the concept of resource adequacy and the future challenges energy planners will have to deal with to guarantee the security of supply due to the increasing dependency of power systems on climate variability. This part of the literature review introduces the role of energy storage as a potential solution to cope with this problem. First, an overview of different types of storage is presented in Section 2.3.1. Long-duration electricity storage and its potential contribution to resource adequacy are explained in Section 2.3.2, while the main technological options for this type of storage are shown in Section 2.3.3.

2.3.1. General Introduction

In the future, a significantly higher reliance on weather-dependent renewables can be expected, alongside expanded storage and greater interdependence with neighboring countries via electricity interconnection. Renewable generation output may drop substantially during certain weather conditions, such as periods of low wind or sunlight. These weather patterns could extend beyond individual countries, impacting neighboring nations as well. While consumers may shift their electricity usage away from such times, there may be limits on how much, or for how long, they can shift. Sufficient additional resources in the resource mix to deliver clean, reliable power at these times i.e. to maintain security of supply and ensure adequacy. Flexibility and resiliency measures are needed to address the fluctuations in supply and demand, which can occur at different timescales from minutes to hours to multiple days [51] as well as across years and decades [52], and ensure reliable and cost-effective power system operations. Flexibility measures can entail supply and reserve sharing, flexible generation, wind and solar generation ratio, demand response, curtailment and electricity storage. The consideration of multiple flexibility options has synergistic and complementary effects. According to [53], power system flexibility types and resources can be categorized at different timescales.

Energy storage can provide a series of services to power systems, including energy arbitrage; transmission and distribution congestion relief; investment deferral; demand shifting and peak reduction; spinning and non-spinning reserves; and seasonal energy shifting. Understanding the characteristics and applications of different energy storage is essential for designing resilient and sustainable energy systems at every timescale. Electricity storage technologies can be categorized based on their discharge duration time, which can vary from a few seconds to several weeks (Figure 2.1) [53]. The relationship between specific power and specific energy investment costs reveals whether storage technologies are more economically viable for short-duration or long-duration applications (Figure 2.2) [53]. Energy storage systems with lower energy investment costs are advantageous for storing large quantities of energy, making them suitable for long-term application. On the other hand, systems with lower power capacity costs encourage a high power demand and short-term usage.

Electrical storage is best suited for small energy capacities and short durations. On the other hand thermal, mechanical, and electrochemical storage technologies are appropriate for medium-sized energy capacities and can handle discharge durations ranging from short to long. Chemical storage is

ideal for large energy capacities and long discharge durations because of the lower cost of storing chemical compounds.

Schmidt et al. [53] organizes the main energy storage technologies commercially available based on discharge duration. The technologies showed in Figure 2.1 are briefly presented below [53]:

- Short-duration storage (seconds to minutes):
 - Supercapacitors: Supercapacitors are ideal for short-duration energy storage needs, such as providing rapid bursts of power to stabilize grid frequency or to smooth out voltage fluctuations.
 - Flywheels: flywheel energy storage can quickly store and release energy in seconds to minutes, making them suitable for grid stabilization and providing backup power during transient disturbances.
- · Medium-duration storage (minutes to hours):
 - Electrochemical battery technologies: they include lithium-ion, sodium-sulfur, lead-acid, and redox flow batteries, and occupy a middle ground between electric and chemical/mechanical storage technologies in terms of discharge duration. These batteries have more balanced power capacity and energy capacity costs, making them suitable for applications requiring a few hours of discharge duration.
- Long-duration storage (days to months):
 - Pumped Hydro Storage (PHS): Pumped hydro storage facilities can store large amounts of energy for extended periods by pumping water to an elevated reservoir during off-peak hours and releasing it to generate electricity during peak demand.
 - Compressed Air Energy Storage (CAES): CAES systems compress air into underground caverns or storage tanks during periods of low electricity demand and expand it through turbines to generate electricity during peak demand periods.
 - Thermal Energy Storage (TES): Thermal energy storage systems store heat or cold energy in materials such as water or molten salts. They can release stored energy over several hours to days, making them suitable for applications such as district heating and cooling, as well as industrial processes.
 - Hydrogen energy storage: hydrogen energy storage involves converting surplus electricity into hydrogen through electrolysis, which can then be stored and converted back into electricity via fuel cells when needed. Hydrogen storage is well-suited for long-duration and seasonal energy storage applications.



Figure 2.1: The effectiveness of different electricity storage technologies depends on factors such as discharge duration and energy capacity [53].



Figure 2.2: Stationary electricity storage technologies are typically categorized based on their power-specific and energy-specific investment costs[53].

This research will focus on long-duration electricity storage, presented in the following sections.

2.3.2. The Design Landscape of Long-Duration Electricity Storage

The term "Long-Duration Energy Storage" (LDES) refers to those technologies that are technically and economically able to store energy for extended periods, typically ranging from days to weeks, and are viable for infrequent use to sustain electricity production over those extended periods [54]. Defining the exact energy capacity range that qualifies as LDES presents challenges. Current literature defines LDES based on an energy-to-power ratio typically falling within the range of 10 to 100 hours or higher [55]. The system value of LDES is still a hot topic of discussion. The increased interest stems from a growing appreciation and acknowledgment of the need for "firm" low-carbon energy resources to com-

plement variable renewable generators like wind and solar. Furthermore, the increasing frequency of extreme weather events such as fires and strong storms is driving the need for the local, clean, affordable, and resilient backup power LDES systems can provide. Batteries are useful for hourly and daily storage because of their relatively low power-capacity costs but do not provide cost-effective seasonal storage due to their high energy-storage capacity cost [56].

For this reason, LDES might be necessary for future power systems to be adequate when demand is higher than power generation for an extended period. Some literature compares the benefits provided by LDES to other forms of flexibility such as short-term storage, interconnection and curtailment. Using a multidecadal economic dispatch simulation Dowling et al. [57] analyses the dispatch patterns of LDES and compare it to one of the short-term storage technologies such as batteries in the U.S. It finds that unlike the significant and seasonal use of long-duration electricity storage for substantial inter-season energy storage, which primarily occurs during summer when solar is most abundant and wind availability decreases, batteries are regularly recharged and discharged in small amount throughout the year. In cases with the greatest displacement of firm generation and the greatest system cost declines due to LDES, optimal storage discharge durations fall between 100 and 650 h (4-27 d) [58]. In a long-term capacity adequacy study in GB, [59] finds that weather patterns extending across North-West Europe and resulting in prolonged periods of low wind during winter do not allow the adequacy of a system that relies uniquely on batteries. Similarly, Cárdenas et al. [60] agree on the fact that different storage technologies will be needed in the UK for different discharge durations to reduce the total electricity cost. Moreover, common findings in the literature show that energy storage plays a more important role as the penetration of VRE increases [61]. [61] points out that when VRE penetration is below 30%, curtailing excess energy is often the most cost-effective solution, as the surplus hours are too few to warrant storage investment. Above 80% a relatively small amount of storage is needed, while when the penetration reaches 100%, there is a substantial increase in the need for storage [61]. With growing VRE shares, the demand for storage increases linearly in terms of power capacity and exponentially in terms of energy capacity [62].

2.3.3. Technological Options for Long-Duration Electricity Storage

LDES encompasses a diverse range of technologies at varying technology-readiness levels and includes electrochemical, chemical, thermal and mechanical options. Among the main technologies that could represent a source of LDSE at a large scale are pumped hydro storage (PHS), compressed air energy storage (CAES), and Power-to-Gas-to-Power (PGP).

Pumped Hydro Storage (PHES)

A pumped hydro energy storage system (PHES) involves two water reservoirs at different elevations, where power is generated as water moves from the upper to the lower reservoir through a turbine during discharge. When electricity demand is low, excess electricity is used to pump water back to the upper reservoir. During periods of high demand, the stored water is released, flowing through turbines that convert the gravitational potential energy into electricity. According to [61], PHES is often the first choice for energy storage due to its maturity, high efficiency (over 80% [53]), low energy costs, long lifespan, and large power capacity. The main factors that could limit this technology are the fact that its geographical location is inflexible, it requires multiple years to be built, has a low energy energy density. Moreover, the creation of water reservoirs could have negative environmental and social impacts. However, according to [61] the effectiveness of PHES may decline in Europe as climate change affects temperatures and river flow, reducing water availability and potentially increasing electricity prices by up to 30% in some regions. Moreover, a higher water footprint is not desirable, due to the increasing pressure on water availability in the years to come.

Adiabatic Compressed Air Energy Storage (CAES)

In a CAES system, electrical energy is used to compress air stored in sealed underground caverns and back-produced when required with energy recovered in a gas turbine. Low specific energy investment costs make CAES suited for long-duration energy storage applications such as storing large amounts of excess renewable electricity supply to be used later. The advantages of this technology are the low specific energy capacity cost and the long lifetime. The main limitation of this technology is the availability of caverns geographically limits cost-efficient underground CAES plants. Another constraint

is the depth range for these caverns, typically falling within the 500 to 1300-meter range. This limitation results from the fact that operational pressure correlates directly with depth, which is limited by the current state-of-the-art technology of power plant components that function optimally within a pressure range of 50 to 100 bar. This presents a less adaptable scenario compared to hydrogen caverns, which can span depths anywhere from 400 to 2400 meters. [63].

Hydrogen

Hydrogen can be used directly as a final energy carrier or can be converted to e.g. methane, synthesis gas, liquid fuels, electricity, or chemicals. The technology that enables the storage of electricity in chemical bonds is the conversion of electricity to hydrogen, also called Power to Gas (P2G). The produced hydrogen can be used directly or can be further processed through methanation with CO2. and the resulting renewable power methane can then be fed into the natural gas grid [64]. Producing electricity from hydrogen can result in slightly improved conversion efficiencies [65]. The main advantage of this technology is the high energy energy density and the potential to use the existing network capacity. However, this can only be provided if underground reservoirs are used to store chemical energy. Suitable formations for underground storage are located in porous rock structures covered with impermeable caprock as well as in man-made caverns in salt rock [64]. At the same time, an additional flexible load can provide benefits to the electricity system being a cornerstone for sector coupling [66]. However, the low round-trip efficiency for re-electrification and the high investment cost for water electrolyzer represent the main barrier, together with the low volumetric energy density and the need for high compression. [61] selects 23 studies that took into consideration power and/or energy system planning problems at usually a national level and with 1-hourly temporal resolution and analyses the techno-economics characteristics of P2G. According to this study, P2G is to be seen as an option to deal with the power surplus rather than a technology to satisfy the current gas demand sustainably. The reason for this is that due to its low efficiency and relative sizes of the electricity and gas sectors, satisfying the gas demand with P2G would require at least three times the current power installed capacity, representing a large cost and providing a large budget that could be used another way to fulfill the same purpose.

The following table summarised the findings discussed above:

Technolog	Category	Advantages	Disadvantages
PHES	Mechanical energy storage	 + Technical maturity + High round-trip efficiency (higher than 80%) + Low specific energy capacity cost + Long lifetime 	 Need for suitable geographical locations Low energy density Long time to build Potential environmental and societal impact of the creation of water reservoirs
CAES	Mechanical energy storage	+ Low specific energy capacity cost + Long lifetime	 Limited by the availability of caverns Low energy density per cubic meter
Hydrogen	Chemical energy storage	 + High energy density + Potential to use existing gas net- works + Potential for sector coupling 	 Low round-trip efficiency (< 40%) High investment cost Need for compression to reach sufficient energy density

Table 2.2: Advantages and disadvantages of the main technologies used for long duration electricity storage

2.4. Current Challenges

As solar and wind capacity continues to expand, and economic incentives and policies are implemented to achieve 100% carbon-free grid goals, long-duration energy storage (LDES) systems could play a crucial role in promoting low-cost carbon-free electricity globally. However, each of the presented technologies has challenges related to their wider deployment. Both PHES and CAES technologies require specific geographical conditions, making their application highly site-specific and constrained to the availability of suitable sites such as heights or salt caverns. P2G suffers from conversion efficiency losses and the high costs of electrolyzers and associated infrastructure. The development of a hydrogen infrastructure for storage, transportation, and utilization remains an obstacle, together with market uncertainties related to a future hydrogen economy.

Moreover, a significant challenge for new stationary electricity systems is the conservative nature of the utility industry. Currently, only PHES and CAES systems are commercially available for bulk storage at utility scale [67]. More advanced technologies, such as electrochemical storage, are still under active development and require further utility-scale demonstration and deployment. The utility industry, which prioritizes low cost, stability, and reliability, is generally risk averse due to concerns over cost, scale-up technologies and lack of proven track record for safe and reliable long-term operation at an industrial scale [10]. Therefore, LDES will need sustained vision, development and investment strategies from both public and private sectors to become integral to a future grid that is affordable, low carbon and resilient.

2.5. Knowledge Gap and Research Question

This research stands at the intersection of resource adequacy, weather uncertainty and long-duration electricity storage.

Resource adequacy revolves around the long-term planning of power systems, ensuring they can meet future demand despite the uncertainties of supply and demand. Energy planners typically assess the power system's resource adequacy using a deterministic or a probabilistic approach and comparing the obtained metrics, such as EENS and LOLE to a reliability standard. While traditional RA assessments focus on dispatchable resources, the rise of renewable energy will cause a predominant increase in non-dispatchable and climate-dependent resources. Current literature highlights the necessity of considering multiple years to yield more accurate results. Moreover, it has identified high-pressure weather conditions, often associated with "low wind, cold snap" scenarios, as a significant source of stress on power systems. Consequently, resource adequacy must recognize weather-related events as key factors for reliability. This is for example done by ENTSO-e in the EERA [45]. However, while energy system modelers have effectively represented the impact of current weather and climate, existing resource adequacy assessments overlook the potential effects of future climate change, which could lead to increased variability in wind capacity and more frequent extreme weather events. For example, only historical data were considered for the EERA 2023 simulations, while a long-term forward-looking climate projection is foreseen from the ERAA 2024 [15].

Furthermore, there is a growing discussion about integrating flexibility options to balance wind and solar energy intermittency. Future power systems are expected to feature a blend of flexible options with complementary characteristics. There remains an ongoing debate about the required extent and form of long-duration electricity storage, compared to other flexibility options such as short-term storage or firm low-carbon energy sources such as nuclear power. Previous studies conducted in the U.S. [57] and in the UK [60] have explored the need for LDES to lower electricity costs and the risk of an energy shortfall, particularly when variable renewable energy sources make up a significant portion of the energy mix. However, no study has investigated LDES's benefits and their main drivers for a future interconnected system in Europe. This leads to the following research question:

What are the capacity value drivers of long-duration electricity storage for an interconnected European power system in 2050 in a context of high penetration of renewables and future climate?

This study aims to conduct energy assessments for an interconnected European system in 2050 where LDES is included in different cases. Weather uncertainty and the influence of anthropogenic climate change will be integrated into the analysis to provide a more accurate reflection of future climate conditions. Through a comparison of case with different levels of LDES, the research seeks to obtain insights into the potential benefits of LDES for future interconnected systems.

This research contributes to advancing knowledge in the fields of energy planning, climate adaptation, and renewable energy integration by addressing key challenges and uncertainties faced by future energy systems. Future researchers and energy planners can rely on this study to make informed decisions on infrastructure investments and optimized resource allocation, ensuring cost-effective solutions for grid reliability and resilience. Policymakers can use these findings to craft policies supporting renewable energy and storage integration in the face of evolving energy and climate challenges.

3

Methods of Approach

This research aims to evaluate the security of supply risk for a European interconnected system in 2050 and the contribution that long-duration electricity storage has in ensuring adequacy. A contemporary adequacy assessment considers uncertain factors within the system, providing a probabilistic measure of adequacy across various feasible realizations of these uncertain variables. Two approaches use probabilistic evaluation: analytical methods and Monte Carlo simulation. The analytical methods represent the system by mathematical models and use direct analytical solutions to evaluate reliability indices from the model [13]. Assumptions are often necessary to simplify the problem, especially when dealing with complex systems [68]. However, this simplification can lead to a loss of significance in the analysis [68] and, as a result, simulation techniques become crucial for accurately evaluating reliability. research uses Monte Carlo simulation, through which reliability indices are estimated by modeling the system's actual random behavior. This chapter provides a theoretical overview of this approach. In particular, energy system modeling is introduced in Section 3.1 as a way to reproduce the power system's behavior and conduct system adequacy assessments through Monte Carlo methods, described in Section 3.2.

3.1. Energy System Modelling

Energy system modeling is a useful mathematical tool that can assist decision-making about the design of the European energy system. The usual way of modeling the evolution of an energy system involves step-by-step optimization over time, typically spanning five to ten years. It starts from an existing energy system state and calculates the most cost-effective setup for the next stage while considering economic and policy constraints, such as potential carbon emission limits. Energy system models enable analysts to build cohesive scenarios depicting how energy is sourced, converted, transported, and utilized and potential shifts in these processes over time. These models have gained renewed significance as they aid in navigating the transformation of energy systems driven by climate policy.

3.1.1. Calliope

This study primarily analyzes a power system model, decoupled from other sectors, constructed using the Calliope framework. Calliope, an open-source tool, facilitates user-friendly modeling of energy systems across various scales, from local city-level systems to fully integrated continental networks. The main target for Calliope is to analyze energy systems that generate electricity from various sources, including numerous renewable technologies [69]. The key elements for Calliope are:

- Carrier: The energy carrier used by a technology, such as electricity, heat, or gas.
- **Technology:** An element capable of managing energy by producing, using, converting or transporting it.
- **Resource:** The origin of a technology's capability to produce, remove, or add energy to the system.

• Location: A node where one or more technologies can be installed. Multiple locations can be included and connected with elements called links.

In Calliope, users can also input resources, energy demand, and lists of locations and technologies. To build an optimization problem, some constraints are set by default, but users can also customize these constraints to fit their specific needs. Following the definition of input variables and constraints, the system solves a linear optimization problem for an entire year, yielding a cost-optimal configuration of the defined energy system. Figure 3.1 shows a possible conceptual scheme to represent how an energy model can be built by using Calliope.



Figure 3.1: Conceptual scheme of Calliope.

In Calliope, there are three modes to run a model: **plan**, **operate**, and **spores**. The default mode is planning, where the model determines the capacities of all technologies. In operational mode, the user specifies the capacities of technologies, and the model manages the energy carrier flows within the system using a receding horizon algorithm. This approach addresses the challenge that in optimization problems, it is often not feasible to know all parameters at a specific time, but only a few days in advance. To simulate this uncertainty, Calliope employs a rolling horizon technique. The model is optimized over the complete time series in smaller parts called scheduling horizons. Each horizon is optimized for a sub-period known as the scheduling window, which is shorter than the horizon. The remaining part of the horizon, not included in the window, acts as a forecast period to provide additional information for the scheduling window. This method ensures that the optimization considers current system conditions and future forecasts.

Calliope has been used in different studies, such as examining national-scale power systems in Britain

[70], South Africa [71] and Europe [72]. Calliope is accessible on GitHub (https://github.com/calliope-project/ calliope [69] and documentation is provided at https://www.callio.pe/.

3.1.2. Modelling Long Duration Energy Storage

Efficiently managing the dispatch of energy storage systems, especially for long-duration and seasonal storage, is complex due to the variability in renewable energy sources like wind and solar. These systems must address short-term fluctuations and extended periods of high and low energy production. Niet [73] highlights that optimization models are essential for balancing storage use and ensuring energy is dispatched at the right times, considering factors like storage capacity, operating costs, and the state of charge. These models must be designed to function across multiple time horizons, from daily balancing to longer-term seasonal shifts. According to [73], one of the main difficulties in modeling energy storage for day-ahead power optimization is the tendency for storage systems to be fully discharged by the conclusion of the optimization model in keeping optimal reserves of storage. [73] describes the main solutions used in industry to model long-duration and seasonal energy storage, which are briefly explained.

Among these solutions, one way to mitigate the issue of depleted storage at the end of the period is to expand the optimization horizon to encompass multiple days, weeks, or even months. According to [73] while this adjustment may enhance the accuracy of models for long-duration energy storage systems, it also complicates the modeling process by adding more variables and equations, which can lead to longer computation times. Alternatively, incorporating a foresight or look-ahead period can help value storage beyond the main optimization timeframe [73]. This method allows the optimization algorithm to access additional information, providing insight into the short-term value of stored energy. Typically, the look-ahead period is modeled with a coarser time resolution or fewer constraints compared to the primary optimization period to keep computational demands reasonable [73].

Another approach to managing long-duration storage is through setting end volume targets. According to [73], this method involves determining predefined energy levels, or state of charge (SOC), which the storage system should aim to reach over various time intervals—whether daily, weekly, or monthly. These targets help maintain storage capacity over time, ensuring that energy is preserved for future needs, particularly for long-duration or seasonal storage scenarios. However, the author points out that one limitation is that this approach might not fully account for operational constraints, such as transmission capacities or the availability of reserves. Additionally, introducing these end-volume targets can increase the complexity of the simulation, leading to longer calculation times[73].

Another practical method for managing energy storage operations is the stored energy value approach, as described by [73]. This strategy assigns a monetary value to the energy stored within the system, effectively determining when to charge or discharge the storage based on market prices. The key principle is straightforward [73]: when energy prices are lower than the assigned value, the system stores energy; when prices exceed this value, the system releases the stored energy. What makes this approach appealing is its simplicity and computational efficiency [73]. By basing decisions on current price signals, the method avoids the complexity of more detailed simulation models. However, as the [73] emphasizes, determining the correct value for the stored energy is critical to the success of this strategy. If the value is set too low, the storage may not be fully utilized, missing opportunities to store cheap energy and discharge during high-price periods. Conversely, setting the value too high could lead to excessive cycling—charging and discharging frequently without yielding sufficient economic benefit. Hence, accurately assessing the stored energy value is essential for balancing operational efficiency and financial returns.

3.1.3. Time Clustering

As explained in Section 2.2.2, wind and solar power generation exhibit substantial year-to-year fluctuations, making it preferable to utilize multi-decade time series data to capture the entire spectrum of this variability. Running an energy system model can be computationally intensive due to the complexity and size of the system, as well as the need to simulate multiple scenarios over long time horizons and consequently to model numerous time steps. The computational burden is often a significant challenge, particularly for high-resolution models that include detailed spatial and temporal data and involve largescale optimization problems. Thus, there is a trade-off between reducing the computational burden and maintaining the accuracy and granularity of the model.

Time resolution reduction methods offer powerful tools to manage the computational burden on energy systems, enabling more efficient and practical simulations without significantly compromising accuracy. These methods aggregate periods with similar characteristics to reduce the number of time steps that need to be modeled. Pfenninger [31] describes three main methods to reduce a model's time resolution and assess their accuracy disparities when analyzing multi-decade time series. The approaches the author presents are downsampling, clustering and heuristic. Downsampling involves reducing the resolution of the entire time series (e.g. from hourly to every 6 hours), while k-means clustering is a common clustering algorithm used to group periods based on their similarity. Heuristic selection involves choosing days or complete calendar weeks based on certain criteria, such as selecting the week containing the maximum or minimum daily average for the given time series or choosing the week where the relative difference between time series is highest or lowest. The results of this research indicate that the different methods yield different outcomes, especially when modeling scenarios with high shares of variable renewable generation. Heuristic methods tend to be more consistent and may be more suitable for scenarios with high levels of variable renewables [31].

3.2. Monte Carlo Methods

Generating system adequacy assessment is used to evaluate the ability of the system generating capacity to satisfy the total system load. Simulation methods, including Monte Carlo techniques, are utilized to replicate the stochastic behavior of power systems. For broad assessments like those conducted at the European level, Monte Carlo simulation is the preferred method. Monte Carlo simulation employs random numbers to represent stochastic simulations, and it can be categorized into two approaches: sequential and non-sequential. In a time-sequential simulation, an artificial timeline is created to depict system components' operational and downtime periods in chronological order [74]. This is done by using random number generators along with the probabilistic distributions of component failure and restoration parameters, producing a sequence of the system's operating and repair cycles [74]. However, as non-sequential methods evaluate system states independently, they cannot account for how energy storage systems charge and discharge and they cannot track how the state of charge (SOC) evolves. For this reason, sequential Monte Carlo is the method that is used in this research.

The required number of simulations is determined to achieve an acceptable level of statistical convergence, calculated based on the standard deviation of the reliability measure [75]. Annual reliability indices such as EENS and LOLE are calculated by averaging the values of ENS and LLD obtained through the simulations until the standard error of the mean meets the selected convergence criteria. The following section outlines the method that was used for reliability evaluation: the state duration sampling approach [75].

3.2.1. State Duration Sampling Approach

The state duration sampling approach involves sampling the probability distribution of the duration of each component's state. The operating state represents when a component is up and available for service, while the repair state represents when it is down and undergoing repairs. This approach aims to generate a chronological system state transition process by combining the chronological state transitions of individual components. In this approach, the state duration distribution functions are utilized, specifically the operating and repair state duration distribution functions, which are usually assumed to be exponential. Since this approach was chosen for the research, it is described in more detail in the following subsections.

Simulation process

In practice, the state duration sampling approach is applied to simulate the up-down cycle of a two-state unit starting from an initial state by sampling the duration of each state. This method creates an artificial operating history for each generating unit, which is then used to evaluate the system's reliability indices over a specified time.

In a two-state component representation, these are the operating and repair state duration distribution functions and are usually assumed to be exponential. Initially, an initial state is allocated to every component and the duration of each component's state is sampled [75]. The duration of a component in its present state is determined using an exponential distribution based on the formula [75]:

$$T_i = -\frac{1}{\lambda_i} \ln U_i \tag{3.1}$$

Where U_i is a uniformly distributed random number between [0,1] corresponding to the i_{th} component; if the present state is the upstate, λ_i is the failure rate of the i_{th} component; if the present state is the down state, λ_i is the repair rate of the i_{th} component.

The chronological transitions of component states are simulated in this way and put together to represent the operational history of the component [75]. This approach uses the component state duration distribution functions.

Two-State Model Representation

A two-state model can represent base load units such as thermal generators, which have long operating cycles while peaking units have short operating cycles. In this case, the generating unit can be either "up" (available for service) or "down" (unavailable for service). When in the upstate, it can transition to the downstate due to a fault. Conversely, from the downstate, it can return to the upstate through repair. The generating unit model provides an artificial operating history of the unit in the state duration sampling simulation by drawing sample values of TTF (Time-to-Failure) and TTR (Time-to-Repair) of the unit. TTF refers to the duration between the start of an operation and the occurrence of a failure, while TTR is the duration taken to restore the system or component to an operating units are exponentially distributed. This means that the failure rate λ of the unit is "memoryless" and constant. The Mean-Time-to-Failure (MTTF) represents the average duration until a base load unit experiences failure, while the Mean-Time-to-Repair (MTTR) indicates the average duration required to restore the unit to an operational state after a failure. MTTF and MTTR are linked to the failure λ and repair rate μ by the following equations [75]:

$$\lambda = 1/MTTF \tag{3.2}$$

$$\mu = 1/MTTR \tag{3.3}$$

Given the MTTF and MTTR of a generating unit, it is possible to sample the TTF and TTR by drawing random variates following the exponential distribution [75]:

$$TTF = -MTTF \ln U \tag{3.4}$$

and

$$TTR = -MTTR\ln U' \tag{3.5}$$

where U and U' are two uniformly distributed random number sequences between [0,1].

An up-down cycle of a two-state unit can be generated starting from an initial state by sampling values of the TTF and TTR. By combining the operating cycles of all units, the total available capacity of the system is obtained.

4

System Modelling and Analysis

After an explanation of the methods used to answer the research question, how these methods were applied in the research is described in this chapter. First, the methodology framework is given in Section 4.1. Then, the power system model that was built as part of this research is explained in detail in Section 4.2, together with the cases that were investigated. Finally, the information regarding the adequacy assessment is shown in Section 4.3.

4.1. Methodology Overview

Employing the Calliope model framework explained in Section 3.1, a linear programming model is constructed to optimize the design and operation of a possible future European power system. The model's objective function aims to identify the design and operation with the lowest total system cost, considering supply capacities, storage capacities, and transmission capacities at each node. The research was divided into two steps both using Calliope to generate different outcomes:

• Step 1: Planning Mode: Initially, a capacity expansion model is utilized to identify the most cost-effective configuration of generation, storage, and transmission investments and operations in 2050. Some of the inputs for this step are obtained by the Ten-Year Network Development Plan (TYNDP) [4]. The capacity expansion is done to incorporate weather variables projected to reflect the climate in 2050, as detailed in Section 4.2.5. and to endogenously obtain the power system's storage requirements. Moreover, this approach allows for the analysis of future peak net demand events across the entire period considered.

This model ensures that national electricity demands, carbon emission targets, and a minimum of 80% share of electricity generated from RES are met. This constraint helps achieve a high share of renewables in the energy mix. It reflects the necessity for significant storage capacity beyond 80%. Moreover, a reserve margin of 0.1 for the peak demand is set. The capacity expansion model uses a greenfield approach and runs hourly simulations for ten years of weather data (1985-1995), accurately capturing the LDES characteristics needed to address seasonal variations in load and VRE generation. In this initial stage, time clustering and averaged dispatch simulations are utilized. This approach facilitates the computational feasibility of modeling a wide range of loads and weather conditions. In this phase of the research, forced outages were not included. The generated design adheres to the technical constraints of available technology components and the political constraints.

Step 2: Operation Mode and Adequacy Assessment After obtaining the design of the power system, the economic dispatch was simulated for different energy storage scenarios, where the storage power capacity and the round-trip efficiencies of LDES and SDES are the same, but the duration of LDES varies from 50 to 200 hours. The simulation is done for all 8760 hours of a year using Calliope's operation mode. Forced outages for thermal generators were included. Monte Carlo simulations were conducted in this mode to obtain the load curtailment time series and calculate the EENS and LOLE for every case.



Figure 4.1 illustrates the main elements of the methodology followed in this research:

Figure 4.1: The model framework employed in this study integrates various input data (depicted in blue), obtained from different sources. Temperature-dependent demand time series were specifically derived using the DESSTinEE tool. Subsequently, an optimization problem was formulated within the Calliope framework, utilizing both supply and demand datasets. The outputs (depicted in green) represent the outcomes generated by the Calliope optimization model. Notably, the installed capacities of battery storage and hydrogen technologies result from the planning mode optimization problem. Additionally, the supply and demand time series are outcomes derived from Calliope's operational mode economic dispatch. to compute the EENS and LOLE metrics. This analysis aids in comprehending the impacts of LDES on the system's adequacy.

4.2. Model Outline

This research focuses on a future European electricity system; This system is represented as a network of nodes, each representing a European country, and the power flows between them. It considers the deployment of renewable electricity supply and storage technologies at each node and the transmission links between nodes while disregarding subordinate network nodes and distribution system power flows. The technologies implemented as sources of electricity include solar power, onshore and offshore wind power, hydro reservoir power, run-of-river hydro, biogas power, nuclear, and gas CCS power plants, batteries, hydrogen, and pumped hydro as a form of storage.

4.2.1. Model Assumptions

This model is a simplified representation of a European interconnected power system that, as any model, is based on a set of assumptions. Below is a list of the main assumptions:

- Cost-driven dispatch decision: Calliope allocates available resources for specified time periods, aiming to minimize overall system costs.
- **Perfect foresight:** It is assumed that renewable energy sources, thermal capacities (including forced outages), grid capacities, and demand are all known in advance with perfect accuracy. This means there are no discrepancies between forecast and actual values, allowing for optimal allocation of storage capacities throughout the year.

- Aggregated demand by study zone: Demand is aggregated at the study zone level, without modeling individual end-users or end-users group.
- Demand is partly correlated to the weather and is unelastic: Demand levels are partially influenced by weather, with temperature changes affecting the use of electrical heating and cooling devices. Demand is considered inelastic to price and remains constant regardless of energy prices.
- **RES production depends on climate:** Solar, wind, and hydropower generation are directly influenced by climate conditions.
- Forced Outages (FO) affect only thermal generators: Nuclear and gas power plants are subject to forced outages, meaning their net generating capacity is not consistently guaranteed.
- **Zonal Model:** The entire power system is divided into zones, with each zone representing a country. The power flow between zones is subject to transmission constraints, representing the maximum transferable electricity between zones.

4.2.2. Input Data

Building the model starts with collecting a large amount of raw input data. The latter is processed to serve as input for the scenario computations. Figure 4.1 illustrates the primary input data gathered and processed for the system's demand and generation.

A lot of the data given in input to Calliope was retrieved from the projections presented by the Ten-Year Network Development Plan (TYNDP) [4]. The TYNDP is a strategic planning document developed by ENTSO-E. It outlines the anticipated development of the European electricity transmission network. The TYNDP serves as a roadmap for identifying and prioritizing infrastructure projects necessary to ensure the reliable and efficient operation of the European electricity system. It considers factors such as integrating renewable energy sources, changes in energy demand, and cross-border electricity flows. The plan is updated every two years to reflect the latest developments and challenges facing the European electricity sector.

The scenarios presented in the report align fully with the goals of the Paris Agreement and the European ambitions for achieving climate neutrality by 2050 remaining neutral in terms of technology and energy carriers. Three scenarios are presented in the report and cover different time horizons:

National Trend scenario (NT)

The National Trend scenario (NT) is in line with National Energy and Climate Policies (NECPs, national long-term strategies, hydrogen strategies, etc.), matched with European targets. Data for electricity and gas in this scenario are sourced from TSOs, reflecting the latest policy and market developments discussed at the national level. The quantification of NT focuses on electricity and gas up to 2040.

Distributed Energy (DE)

In addition to the NT, which is aligned with national policies, ENTSO-E has developed two COP21compliant scenarios, both aiming to reach the 1.5°C target of the Paris Agreement. The Distributed Energy (DE) scenario outlines a trajectory toward achieving carbon neutrality for the EU27 by 2050, along with a minimum 55% reduction in emissions by 2030. This scenario is driven by the willingness of society to reach energy self-sufficiency using readily accessible indigenous renewable energy sources. It involves a shift in lifestyle and a significant decentralized effort to decarbonize through local initiatives by individuals, communities and businesses, with support from authorities. This approach aims to maximize renewable energy production within Europe and substantially reduce reliance on energy imports. This is obtained by achieving a reduced energy demand through circularity and better energy consumption behavior and a focus on decentralized technologies (PV, batteries, etc.) and smart charging, and a minimal integration of nuclear power and Carbon Capture and Storage (CCS).

Global Ambition (GA)

The Global Ambition (GA) scenario outlines a route towards achieving carbon neutrality by 2050, along with a minimum 55% reduction in emissions by 2030, propelled by a global shift towards meeting the targets set in the Paris Agreement. This scenario includes the development of a wide array of renewable

and low-carbon technologies, many of which are centralized, and emphasizes the utilization of global energy trade as a means to accelerate decarbonization efforts. Economies of scale are leveraged to achieve significant cost reductions in emerging technologies like offshore wind, and the scenario also considers importing decarbonized energy from competitive sources as a viable option. In this scenario, energy demand also declines, but priority is given to decarbonization of the energy supply, focusing on large-scale technologies and the integration of nuclear and CCS.

The model constructed with Calliope does not account for decentralized technologies or smart charging; rather, it prioritizes the deployment of large-scale facilities. Additionally, it involves importing decarbonized energy from external sources, reflecting a broader, global approach to decarbonization. Due to these characteristics, the GA scenario was considered more aligned with the objectives of this research, leading to the selection of input data corresponding to this scenario.

4.2.3. Geographic Scope, Transmission Grid and System Scale

An interconnected power system was modeled for selected European countries, with the choice of countries limited by the availability of weather data that could represent future climate conditions, as detailed in Sections 2.2.3 and 4.2.5. The study area entails 8 locations (Great Britain, Ireland, France, Germany, Denmark, Austria, Belgium, Netherlands) each of which is considered to be a transmission network node. However, the model does also not account for the existing topology of the transmission system. The transmission grid is modeled as direct net transfer capacities between network nodes. The TYNDP GA scenario gives the maximum export capacity between nodes.



Figure 4.2: Transmission grid of the model. Each location is considered a network node and the transmission grid is modeled as a direct transfer between them.

4.2.4. Time Resolution and Horizon

For planning mode, the model uses hourly time series from 1985 to 1995 for weather variables and demand. This choice balances the need to consider multi-decade datasets to account for renewable energy variability and the need to limit the computational burden. The dataset contains "normal" weather years and particularly cold years for the geographical scope considered (1985, 1987, 1993, 1995) [30]. This approach ensures that both average and extreme weather conditions are considered when building generation portfolios. More years could have been included but this would have resulted in a larger problem size and more computational time required to solve the optimization.

To reduce the model's time resolution computational time, a heuristic model is adopted as in [31]. Using Calliope, one can apply one or more time masks, retaining parts of the time series at maximum resolution (unmasked) and reducing the resolution of other parts (masked). It is also possible to adjust the temporal scale of the extremes that are analyzed, influencing both the granularity and the type of insights that can be derived from the data, and to add symmetric padding (a fixed duration before or after the extreme values). While daily extremes can be useful for identifying the most demanding days and planning for daily resource adequacy, finding weekly extremes can be beneficial for understanding weekly patterns and planning for longer-term resource needs. Using 10 years of simulated wind and photovoltaic (PV) generation data, the model is designed to identify five periods, each lasting three weeks, that represent the extremes of wind and PV generation. This choice is based on current research highlighting significant inter-annual variability in renewable power output [76]. Not every year experiences extreme weather conditions, as evidenced by wind droughts with a one-week duration having a return period of about two years, and those lasting two weeks occurring every five years [77]. By selecting five distinct three-week periods of minimal wind power output, the aim is to include these critical events and effectively capture the dynamics of LDES. This choice also balances computational efficiency, as increasing the number of periods would require handling more timesteps. The resolution of the remaining time series is then reduced to 6-hour intervals. This approach facilitates the long-term planning process.

For the operational mode, an hourly resolution for a one-year simulation is used. This approach enables to capture of diurnal patterns, accounts for renewable energy variability, and optimizes the generation and storage of resources. The operation of the modeled power system is optimized for a limited time horizon, called a "rolling horizon", taking into account a certain number of hours, usually called "horizon". In this case, decisions regarding operations for each 24 hours are determined by optimizing over 72-hour intervals. This procedure reduces computational burden by segmenting a large optimization problem into smaller ones and it better replicates the actual operation of the power system [18].

4.2.5. Generation

Wind and Solar Power

The available wind and solar energy is calculated based on the installed capacities of the reported technologies and wind and solar load factor profiles for each country. Future capacities installed in every country were retrieved from the projections given by the TYNDP GA scenario [78]. These capacities are given in input to Calliope as the maximum installable capacity for the optimization problem. As explained in Section 2.2.2, wind and PV capacity factors must reflect the future climate, taking into account potential variations due to anthropogenic change. [2] provides a collection of hourly times series relating to future climate projections through delta calibration. The advantage of this calibration lies in its direct adjustment of past weather events that have already been experienced. This process reduces the uncertainty associated with the variability of GMCs output, simplifying the assessment of climate change's impact on past weather patterns. In this way, it is possible to assess what impact past events could have in the future on the power system. The datasets provide hourly data on wind speed, solar irradiation, and temperature, together with the corresponding wind and PV capacity factors. These capacity factors were estimated by using physical models that reproduced the onshore and offshore wind power curve and the PV's power curve. These models are explained in [2] together with their performance assessment. The following subsection briefly explains the concept of delta calibration followed by [2] for the creation of the timeseries.

Integrating Historical Data for Future Scenarios

First, historical meteorological data was retrieved from the ERA5 reanalysis from 1950 to 2020 available on Copernicus Climate Data Centre [46] to create this set. ERA5 is the fifth generation of atmospheric reanalysis to be produced at the European Centre for Medium-Range Weather Forecasts (ECMWF) and currently covers the period from 1950 to the present. The dataset has a spatial resolution of 30 Km and consists of hourly 2-meter temperature, accumulation of surface solar irradiance, and hourly 10 and 100 m wind speeds. The latter was consequently bias-corrected by the author [2] to adjust the magnitude of the ERA5 wind speeds to those available from the Global Wind Atlas dataset [79]. Second, 5 climate model simulations that offered hourly climate variable output were selected and, having a higher spatial resolution than the ERA5 data interpolated onto the ERA5 grid before calibration. The climate models that were used are:
Model name	Modelling centre			
MOHC-MM-1hr (ens2)	UK Met Office			
MOHC-MM-1hr (ens3)	UK Met Office			
MOHC-HH-3hr	UK Met Office			
EC-EARTH3P-HR	European Community			
EC-EARTH3P	European Community			

Table 4.1: Climate models used by [2]. Data retrieved from [3].

The climate model calibration that was used is known as delta correction. Delta correction is a way to fix any mistakes or biases in climate model data by comparing it to actual weather observations. Instead of changing the climate model data directly, a model is used to predict how climate change might affect weather patterns, and then adjust historical observations based on these predictions [2]. The advantage of this type of calibration is that it directly adjusts past weather events that are already been experienced, attenuating the uncertainty related to the variability of GMCs' output. This makes it simpler to see how climate change might be affecting the weather experienced in the past. From an energy planner's point of view, this information can be used to understand how climate change might have impacted past extreme weather events, giving them insights into how to plan for similar events in the future and make the energy systems more resilient. Figures 4.3, 4.4 and 4.5 show a comparison for the Netherlands of weather variables such as temperature, wind and solar capacity factor between the historical data (ERA5) and the GCMs presented in Table 4.1.



Figure 4.3: Historical and delta-calibrated average daily temperature of Netherlands for 1985. It can be noticed that temperatures are higher for the calibrated series, as expected due to the effect of climate change. Day 0 corresponds to 1st June 1985.



Figure 4.4: Historical and delta-calibrated daily solar capacity factor of Netherlands for 1985. The conversion from 2 m temperature and incoming surface solar irradiance to solar power capacity factor is explained and retrieved in [2]. Calibrated data appears to be on average higher than historical data. Day 0 corresponds to 1st June 1985.



Figure 4.5: Historical and delta-calibrated daily offshore wind capacity factor of Netherlands for 1985. The conversion from 100 m wind speeds into wind power capacity factors through offshore wind power curves is retrieved from [2]. Calibrated data appears to be on average than historical data. Day 0 corresponds to 1st June 1985.

Biofuel

The primary sources of biomass for energy production are agriculture, forestry and waste. Agriculture sector energy sources are considered energy crops and residues (primary, secondary and solid). "Energy crops" refers to the crops cultivated primarily for the production of end-use energy carriers, such as sugar, starchy and oily crops. Residues include dry and wet manure from cattle, suitable for gasification, olive pits and the waste obtained from pruning of permanent crops. Estimations of biomass

potential for the year 2050 were retrieved from [80]. Based on land use, agricultural practices and proceeded areas, a low bioenergy availability scenario was assumed, therefore excluding dedicated farming for energy crops and focusing instead on residuals and wastes. The model does not differentiate between materials derived from various feedstocks, assuming a uniform combustion efficiency of 45% for all biomass sources.

Nuclear and Gas CCS

Nuclear and gas CCS technologies are likely to be considered as potential components of the energy mix due to their capacity to generate low-carbon electricity. Nuclear power is a well-established low-carbon energy source that has been part of the European energy mix for several decades. While there are concerns regarding nuclear waste disposal and safety, nuclear power plants produce minimal greenhouse gas emissions during operation and can provide a stable baseload power supply. Gas CCS involves capturing carbon dioxide emissions from natural gas power plants, and storing them underground to prevent their release into the atmosphere. Installed capacities for these technologies are retrieved from the TYNDP GA scenario. Efficiencies for nuclear power and gas CCS are set respectively at 033 and 0.53.

Hydro Run of River and Reservoirs

While hydro run-of-river power plants are modeled as a continuous supply of electricity with no storage facility, reservoirs are modeled with internal storage that provides both base load as well as the ability to be shut down and started up at short notice according to the demands of the system (peak load). The assumption that was taken is that the potentials for hydro run-of-river and hydro reservoirs are currently largely tapped, with minimal expansion potential. Therefore, hydro-generation capacities are limited to today's levels. Hydropower installed power and energy capacity data was retrieved from the JRC Hydro Power Database was utilized [81] together with the values presented in Euro-Calliope [72]. Timeseries of capacity factors per unit installed capacity for hydropower reservoirs and run-of-river hydropower plants in the studied for 30 climatic years (1981-2010) are retrieved from [82].

Pumped Storage Hydro

Similar to hydro run-of-river and hydro reservoir capacities, pumped storage hydro capacities are assumed to be largely tapped and do not allow for capacity expansion. Power and storage capacity are determined using the JRC Hydro Power Database [81] and Euro-Calliope [72]. Therefore, today's pumped hydropower and storage capacity deployment is limited. A round-trip efficiency of 78/% is assumed. Furthermore, it is assumed that enough storage options are available to ensure that the only limit is the technical potential, without additional time constraints.

4.2.6. Short-Term and Long-Term Storage

Short-term and long-term storage are assumed to be deployable in all the countries. Short-term storage is modeled as Lithium-ion batteries, while long-term storage is represented by P2G, as these technologies are expected to dominate their respective applications [83]. In its Calliope European model, Tröndle et al. [72] models short-term and long-term storage using two technical parameters: the ratio between power and storage capacity and the round-trip efficiency. This approach is followed also in this research. A round-trip efficiency of 86% is assumed for short-term storage and 40% for long-term storage [72].

In planning mode, short-term storage is limited to a maximum storage capacity of 10 hours at full power [84], while long-term storage's discharge duration is constrained to be higher than 10 hours. The capacity expansion is shaped by the assigned costs for each technology's installed capacity and energy storage. In this case, LDES has a higher power capacity cost but a lower storage capacity cost compared to the ones of the battery. These costs were retrieved by the sources cited in Section 4.2.6 and reflect the economics of storage explained in Section 2.3.1.

In operational mode, short-term storage discharge duration, also described by the energy-to-power (E2P) ratio, is fixed at 10 hours, while the E2P power ratio of LDES is set at 50, 100 and 200 hours. This is done to study the effects that storage duration has on power systems' resource adequacy. Moreover, to compare SDES and LDES based on reliability metrics such as EENS and LOLE, the analysis examines how varying the power and energy capacities of both storage types impacts these metrics. The studied cases are presented in Chapter 5.2.

Technology costs and parameters

Cost estimates for the year 2050 primarily rely on [85] [86] [83]. The cost of the technology is calculated by combining the capacity cost, annual maintenance cost proportional to the installed capacity, and the variable cost for each unit of electricity generated. For all hydropower technologies, only annual maintenance and variable costs are considered as done in [72], as it is assumed that maximum capacities are already built today, so the capacity cost of hydropower does not affect the results. Technology life-time and an interest rate of 5% are used for all technologies to calculate the corresponding Levelized Cost Of Electricity (LCOE).

4.2.7. Electrical Load

Demand profiles for future years were synthesized using DESSTinEE (Demand for Energy Services, Supply and Transmission in EuropE) [87]. DESSTinEE is a model designed to simulate the European energy sector in the year 2050, going from the demand for energy services to the hourly patterns of electricity demand and generation. It is important to note that this model serves as a tool for examining pre-existing or user-generated scenarios for the future; it does not predict future outcomes.

The modeling process starts by taking the annual demand of each country and building up hourly profile sector by sector. The annual demand is obtained from the TYNDP GA scenario retrieved from [88]. Each country's load curve is divided into key economic sectors, including residential, commercial, agriculture, industrial, road and rail. Within each sector, built environments are further segmented into space heating, water heating, cooling and other appliances. Daily profiles comprising 24 periods are defined for each sector and end-use, incorporating variations for summer/winter and weekday/weekend patterns to reflect fluctuations in human and economic activity. The model generates annual profiles spanning 8760 periods for each sector, with space heating demands adjusted based on the number of Heating Degree Days (HDD) experienced. The temperature that is given in input is the same that is used to model the power generation, so the demand is temperature dependent, as shown in Figure 4.6. Each profile has a mean and standard variation for 24 hourly values. Stochastic variation is introduced to each profile by randomly fluctuating within this range to simulate the inherent variability of human behavior. and the sector profiles are aggregated to derive the national load profile. Together with the input that is given to DESSTINEE, this framework captures four main factors shaping demand: fluctuations in electric heating demand driven by temperature variations, shifts in the technology mix for heating and transportation, sectoral changes such as improvements in efficiency, and year-to-year variations in weather conditions.



Figure 4.6: Historical and delta-calibrated demand timeseries of Netherlands for the year 1985. As the temperature is higher for the different GMCs, so is the demand needed for cooling in the summer period Day 0 corresponds to the first hour of the year.

Load Curtailment

Calliope allows the creation of a variable of unmet demand in the optimization to capture any mismatch between supply and demand. This variable is assigned a very high cost, ensuring it is only utilized when necessary. However, load curtailment is implemented in the model as a supply technology with a variable cost corresponding to the Value Of Lost Load (VOLL). This approach allows for more flexibility, as this cost can be easily adjusted.

VOLL is affected by several factors, such as the type of customers, geographic location, timing, severity, and duration of power outages [89]. According to [18], load shedding should be implemented in "tranches" to reflect the increasing VOLL with higher amounts of load shedding. By curtailing the load in tranches, the socio-economic costs of load shedding can be minimized [18]. Additionally, this differentiated approach to VOLL allows to uniquely define the operation of energy storage during periods of scarcity, as demonstrated by [18]. Various studies report that estimates for the VoLL range from approximately $1500 \notin$ /MWh to $22940 \notin$ /MWh [90]. Therefore, a primary load-shedding technology with a cost of $9000 \notin$ /MWh is created for the planning mode. Additionally, a secondary load-shedding variable is created with a cost of $10000 \notin$ /MWh. The capacity of the primary load-shedding technology in each country is set slightly below the electricity peak demand of the respective country.

Regarding the operational mode, Calliope's optimization aims to minimize costs. However, in certain optimization scenarios, this can lead to higher spikes in load curtailment during extreme events or system stress conditions. To balance properly the energy shortfalls among multiple countries, the model implements multiple tranches with increasing minimal cost differences (approximately 1%). Each tranche's capacity is allocated based on proportional demand across countries, typically in increments of 10% of each country's peak demand. This allocation structure means that the first tranche can curtail 10% of a country's peak demand, with subsequent tranches curbing larger proportions, thereby ensuring a proportional and equitable distribution of curtailment capabilities relative to each country's total consumption. Fig. 4.7 shows Calliope's optimized dispatch with and without load-shedding tranches. Implementing multiple load curtailment tranches results in more frequent curtailment events with lower peaks. This approach doesn't impact the total EENS, but it can influence other adequacy metrics like LOLE.

This approach assumes that proportional curtailment based on peak demand is a fair method for managing load shedding across different countries, ensuring that no single country is disproportionately affected. Additionally, the model uses a value of VOLL that is the same for every end user, without accounting for variations that might exist depending on the purpose of the electricity demand. This is a simplification, as research has shown that VOLL varies across several factors, such as customer class, location, timing, frequency, magnitude, and duration of power outages [89]. Determining the appropriate VOLL values is a complex field of research [91] that requires further attention, however, the approach used here can provide a general indication of when and to what extent load curtailment occurs during the simulation.



(a) Load curtailment distribution when only the tranches of the planning mode are used for the simulation.



(b) Load curtailment distribution when tranches are implemented. The spikes are lower as the curtailment is more evenly spread. The legend shows different load-shedding tranches, each represented by a distinct color, arranged from least to most expensive. Multiple tranches can be activated simultaneously once the tranche immediately before, and therefore less expensive, has reached its maximum load curtailment accessible.



(c) Load curtailment distribution detail when tranches are implemented.

Figure 4.7: Load curtailment implementation in Calliope's operational mode. Using multiple load curtailment tranches allows for lower peaks but a more frequent curtailment. Therefore this approach does not affect EENS but can affect other adequacy metrics such as LOLE. In fact, while the total amount of energy curtailed might remain the same, the frequency of curtailment events increases. This reflects that LOLE indicates the likelihood of experiencing load-shedding events but does not account for the severity of each event.

4.3. Adequacy Assessment

This study will be evaluated through the lens of generation adequacy assessment, simplifying the representation of the transmission grid. RA assessments are typically conducted over multiple years, as seen in the ERAA [15], to capture the inherent variability in weather that affects VRES and demand. In the planning phase of this research, multiple years of weather data were used to account for uncertainties in VRE output and demand due to annual weather fluctuations. However, for the RA assessment, only a single year is selected. This year is chosen to evaluate the impact of LDES on system adequacy under a worst-case scenario. As a result, the calculated EENS is a year-specific metric. This approach reduces the number of simulations required for statistical for statistical convergence. While focusing on a specific weather year can offer deeper insights into how weather influences periods of power system stress [92], it also introduces less uncertainty into the model by not considering all available weather years.

The methodology that was followed in this research uses sequential Monte Carlo simulations to model individual unit-level thermal outage states, as explained in Section 3.2.1. The scenarios used for the assessment are obtained considering different asset-forced outage (FO) occurrences for thermal and nuclear generators. These outages are simulated using a two-state Markov model (explained in Section 3.2.1) under the assumption that outages occur independently. In contrast, the availability of VRE sources is not derived from MC simulations but is based on the weather data presented in Section 4.2.5. By combining random outage scenarios, a diverse set of potential system states is generated for the chosen year. This approach enables a probabilistic assessment of results, which is highly suitable for the dynamic nature of modern power systems.

4.3.1. Selection of Climate Year

First, a specific climate year is chosen to assess the power system. The period that was chosen goes from the 1st of June 1984 to the 1st of June 1985. The winter period in 1985 was particularly relevant because it presented a period of extreme weather conditions that significantly impacted energy demand and supply [30]. 1985 experienced one of the coldest winters in recent European history, leading to exceptionally high electricity and heating demand. The cold conditions also affected hydropower generation, as lower temperatures can impact water flow and reservoir levels [93]. The year 1985 serves as a notable example of two distinct weather phenomena depicted in Figure 4.8: a widespread cold spell across Northern-Western Europe affecting various countries, and the 'dunkelflaute,' an unusually prolonged period of low wind speeds exacerbated by a persistent high-pressure anticyclone over the North Sea. By referencing 1985, it is possible to gain valuable insights into the role of long-duration electricity storage during stress on the power system induced by extreme weather events. For this year, a dataset was created following the same approach explained in Section 4.2, containing the following data for every country in the scope area:

- · Temperature-dependent demand time series
- · Wind capacity factor time series
- · Solar capacity factor time series
- · Run-of-river and reservoir hydropower plants capacity factor

An example is shown in Figure 4.9. To prepare for potential critical events anticipated in the winter of 1985, the simulation begins in the summer of 1984. This early start allows for the charging of storage systems, ensuring they are ready to provide electricity during periods of potential adequacy risks.



(a) Temperature map of 15th January.



(b) Pressure map of 21st February. Dunkelflaute and cold snaps are usually associated with high-pressure events.

Figure 4.8: Extreme meteorological conditions of 1985. Source: [1]



Figure 4.9: Onshore and offshore wind capacity factor timeseries. It can be noticed that the capacity factor remains below 0.2 for onshore and 0.4 for offshore for several days.

4.3.2. Forced Outage Profiles

As a second step, multiple sets of random FO occurrences, represented by hourly time series, are generated for this year. To model the operating history of a generating unit together with its up-downup cycles, parameters such as MTTR and MTTF are needed. [4] provides the Forced Outage Ratio (FOR) and the MTTR that was used for EERA 2023 to model forced outages for nuclear and gas generators. FOR represents the likelihood of a forced outage and can be described by the following equation [68]:

Unavailability (FOR) =
$$U = \frac{\lambda}{\lambda + \mu}$$
: (4.1)

where λ is the expected failure rate and μ is the expected repair rate. Combining equation 4.1 with equations 3.2 and 3.3 it is therefore possible to sample TTF and TTR by using the method described in Section 3.2.1. Table 4.2 shows the with calculated parameters:

Table 4.2: Parameters used to generate the operating history of nuclear and gas power plants. FOR and MTTF were retrieved from [4], while the rest was calculated using the equations presented above.

Parameter	Value		
FOR	5.6 %		
MTTR	24 hr		
μ	0.04167 hr ⁻¹		
MTTF	404.85 hr		
λ	0.00247 hr ⁻¹		

Every country presents a certain installed capacity of nuclear and gas power plants, given by the generation of the reference scenario, presented in Section 5.1. To create different generating units, the total installed capacity of nuclear and thermal power was divided by respectively the average size of big nuclear and gas power plants. A random initial state of up or down and forced outage patterns were created for each of these units. The figure below shows an example of the up-down-up cycles of one of these units:



Figure 4.10: Operating history of a generating unit created for the selected year at an hourly resolution. If the unit is operating, a value of 1 is attributed to the state of the unit, while it is undergoing an outage, a value of 0 is given to the state. The initial state is randomized for each generating unit of the model.

The number of forced outage samples used for the adequacy assessment was determined only after the model reached convergence, as explained in Section 4.3.4. Each model run is performed for the selected year and includes a single instance of a randomly generated forced outage sample. Combining the selected year and M FO samples for this year yields a total of M model runs, for which it is possible to calculate probabilistic resource adequacy metrics such as EENS. Each of these model runs is optimized individually.

4.3.3. Storage Model Configurations

According to [73], the challenge of scheduling storage dispatch over time is equally complex for both long-duration and seasonal storage systems. Section 3.1.2 explains the main methods that are used to model LDES dispatch in literature. As explained in this section, an incentive needs to be given to prevent hydrogen from being depleted by the conclusion of every optimization horizon. As every approach has its own benefits and disadvantages, different strategies are implemented and evaluated in this research:

- Discharging is minimized by assigning a high cost to electricity production from LDES, with a foresight window of 72 hours.
- Discharging is minimized by assigning a high cost to electricity production from LDES, with a foresight window of 200 hours.
- Discharging is discouraged by imposing a high production cost for LDES, while LDES and battery charging are incentivized by assigning a negative cost to the electricity used for charging.

A high production cost is assigned to LDES to ensure that, in the merit order, it is the last technology to be dispatched, only being utilized during times of energy shortages. To achieve this, the production cost is set higher than that of other generators and storage technologies, but still lower than the cost of shedding load. Specifically, since the first tranche of load shedding is priced at 10,000 \in /MWh, LDES is assigned a production cost of 1,000 \in /MWh. This positions hydrogen as a peaking resource, deployed when net demand is high. In the final strategy, battery and LDES charging are incentivized by assigning a negative cost to the electricity consumed during charging. This effectively means these technologies earn revenue for charging, encouraging their use. The negative cost must be small enough to prevent excessive charging, but large enough to have an impact. After multiple tests, it was found that a minimum value of around 1e-10 \notin /MWh is required to trigger this effect. Since batteries are needed for daily electricity supply, their charging is prioritized over LDES. As a result, the negative charging cost is set at 0.9 \notin /MWh for batteries and 0.5 \notin /MWh for hydrogen.

4.3.4. Calculation of EENS

By aggregating the load-shedding tranches that are obtained from the dispatch, it is possible to calculate EENS by using Equation 2.1. The standard deviation that is associated is given by:

$$\sigma^{2} = \frac{1}{N(N-1)} \sum_{i=1}^{N} \left[ENS_{i} - EENS \right]^{2}$$
(4.2)

where *N* is the number of runs and ENS_i is the sample value of the i_{th} run. The errors that are assigned to the calculated EENS values are given by the formula for a 95 % confidence interval:

Confidence Interval =
$$EENS \pm z \left(\frac{\sigma}{\sqrt{N}}\right)$$
 (4.3)

Where *N* is the total number of samples, σ , is the standard deviation, and z = 1.96 is the z-score corresponding to a 95% confidence level.

FOs can affect model outcomes differently depending on the specific demand and supply conditions. For instance, a major power plant experiencing an FO may pose significant adequacy risks in highdemand and low-renewable energy production scenarios. At the same time, in scenarios with high renewable energy production, the impact of such an FO may be less significant. Therefore, the results of model runs can differ significantly based on these factors. To ensure the reliability of the outcomes, the influence of additional MC simulations on the results should be minimal. The simulation should be concluded once the estimated reliability indices reach a predetermined confidence level. The purpose of implementing a stopping rule is to strike a balance between the required level of accuracy and the computational effort needed. The coefficient of variation derived from the ENS on the entire geographical scope is used as the convergence criterion in the Monte Carlo simulation. It is defined as:

$$\alpha = \sigma/E(X) \tag{4.4}$$

where E(X) is the estimated expectation of ENS and σ , is the standard deviation.

The stopping rule used is the following: the simulation stops after reaching a predetermined number of samples, at which point the coefficient of variation is assessed for acceptability. If the coefficient of variation does not meet the desired criteria, the number of samples can be increased to improve accuracy. The coefficient of variation α is kept below 2%.



Results

This chapter introduces the key results of this research, focusing on the performance of LDES in ensuring resource adequacy. It will analyze the generation and storage capacity expansion results in Section 5.1, the impact of different LDES dispatch modeling approaches in Section 5.2, and how variations in LDES discharge duration affect the system's ability to meet electricity demand in Section 5.3. Moreover, a comparison of the performance of LDES with SDES in different scenarios is explained in Section 5.4. Through this chapter, the aim is to understand the role of LDES in future power systems, especially in high-renewable penetration contexts.

5.1. Generation and Storage Capacity Expansion

This section presents the results obtained through the first stage of this research, the planning mode, for which a period of ten years (1985-1995) was considered.

Figure 5.1 presents the total generation capacities for all renewables and fossil-fired technologies projected for 2050. The countries with the highest energy capacities are France, Germany and Great Britain. The total VRE capacity is projected to be 1225 GW, primarily from photovoltaic (546 GW), offshore wind (353 GW), and onshore wind (305 GW). Gas CCUS contributes to the generation mix by adding 57 GW. In terms of annual electricity generation, offshore wind is expected to produce 690 TWh, PV 637 TWh and onshore wind 598 TWh, representing 33%, 30 % and 28 % of the total gross electricity generation, respectively. The significant share of electricity generation from photovoltaic, onshore wind, and offshore wind sources in this scenario reflects the VRE (wind, solar and hydropower) constraint of 80% imposed in Calliope for this mode. This constraint is designed to create a system with a high proportion of renewable energy, as needed in this research. It accounts for the fact that exceeding 80% would require a substantial increase in storage capacity, as discussed in Section 2.3.2. Specifically, VREs account for of 91% power generation (1826 TWh per year), while dispatchable technologies, including biomass, nuclear, conventional hydroelectricity and gas contribute 9% (180 Twh).

It is important to highlight that VRE sources, such as wind and solar, fully utilize their maximum allowed capacity in the model's capacity expansion constraint. In contrast, other technologies, such as nuclear and gas power, do not reach their maximum potential. For example, in France, nuclear power only achieves an installed capacity of 2 GW, despite a model limit of 45 GW. This disparity may be due on one side to the prioritization of VREs and on the other side, to the greenfield approach used in the system's design. The capacity expansion model prioritizes VRE installation, driven by constraints favoring renewable energy. At the same time the greenfield approach, which designs the energy system from scratch without accounting for existing infrastructure, inherently disadvantages technologies like nuclear and gas. These technologies have higher capital costs compared to VREs and storage, making them less economically attractive.

The capacity expansion model allows for an endogenous determination of the storage capacity needed for the system, particularly hydrogen storage. This is especially challenging to calculate since hydro-

gen is often linked to other sectors, such as industry and transportation, complicating its role in the electricity system. Figure 5.2 illustrates the total installed energy storage power capacity for all storage options across the geographical scope. The capacity expansion is influenced by the cost that is given to each technology in terms of installed capacity and energy storage. The endogenously determined total storage power capacity amounts to 111 GW. Having the lowest installed capacity cost, the largest share comes from lithium-ion batteries, which provide 61 GW, while hydrogen storage contributes 30 GW. While the quantity, technology-specific composition, and spatial distribution of storage power vary widely across the observation area. Batteries dominate storage energy capacity for most countries but Austria, where a considerable amount of PHES is available. This is because while it was assumed that there is unlimited storage energy capacity potential for lithium-ion batteries and hydrogen storage, the technical potential of PHES can influence the storage portfolio. The optimization prioritizes the expansion of cost-effective technologies, such as PHES. When additional storage is needed to meet the 80% VRE constraint, the model will continue expanding these technologies until their technical potential is exhausted. Nearly all model regions have maximized their PHES potential. The existing capacities of PHES systems, which amount to 20 GW, are included, assuming they will not be retired by 2050 due to the long lifespan of water reservoirs. The distribution of pumped hydropower plants also reflects the availability of this technology in the territory, which depends on the altitude.

As explained in Section 4.2.6, this stage of the research required setting the discharge duration for batteries to be less than 10 hours, while hydrogen systems had a discharge duration exceeding 10 hours. In Section 5.2, the effect of adjusting the discharge duration of hydrogen, also known as the energy-to-power ratio (E2P), on EENS will be examined.



Figure 5.1: Technology-specific installed generation capacities for Europe in 2050. VRES generate 91% of the total power output, while dispatchable technologies such as biomass, nuclear, conventional, hydroelectricity and gas provide the remaining 9%.



Figure 5.2: Technology-specific storage power for Europe in 205. Hydrogen accounts for the majority of energy storage capacity in most countries, except for Austria, where a significant amount of PHES is available.

Figure 5.3 shows the new transmission links and their capacities. There is considerable transmission capacity between Austria, Germany, and France and Great Britain. Germany and Great Britain also act as major hubs in the network, playing a central role in moving electricity around the region. These values correspond to the maximum installed capacities provided as input, based on projections from the TYNDP for 2050. [4].



Figure 5.3: AC transmission in 2050. Germany and Great Britain serve as key hubs in the electricity network, facilitating power transmission across the region. Significant transmission capacity exists between Austria, Germany, France, and Great Britain, with installed capacities reflecting TYNDP 2050 projections [4].

Moreover, an analysis of net peak demand is conducted. Studying net peak demand hours is important for understanding the capacity requirements and operational strategies needed for long-duration electricity storage, as it helps reveal when storage systems must discharge electricity to meet high demand. Figure 5.4 shows the seasonal peak net load frequency that was done for the top 10 net load hours in each country of the scope, while Figure 5.5 shows the frequency for different years of the same events. In most countries, the top net load hours correlate to peak load hours. Within the studied period, each year contributes to at least one of the top 10 peak net load events across the entire scope of the research, whereas every country experiences at least four distinct years with the highest peak net demand hours. Given that peak net load hours occur over multiple weather years, relying on data from only one weather year when planning capacity expansions is likely to result in a system that lacks adequate capacity during operation in different weather conditions. Therefore, considering a broader range of weather data is important to ensure sufficient capacity that can meet varying demands across different years.

Across the weather years examined, the majority of the highest peak net load hours are concentrated in the winter and autumn seasons. Significant shifts in electricity demand patterns, such as increased electrification, could intensify the trend towards winter peak hours. Interestingly, none of the countries present peak net load hours in the summer season. Since summer typically lacks peak net load hours, generator maintenance and downtime will probably occur during this period. The assumption that climate change will lead to greater summer electricity needs is valid, but the risks associated with resource adequacy remain predominantly concentrated in the colder months, where demand spikes are driven by heating and reduced renewable generation. Consequently, planning for resource adequacy still prioritizes the winter season as the primary period for ensuring the reliability of power systems.



Figure 5.4: Seasonal occurrence of the top 10 peak net load hours for the resource assessment region considered. The peak net load hours are predominantly concentrated in the winter and autumn seasons across the analyzed weather years. Increasing electrification may further amplify the trend toward winter peaks. Notably, no peak net load hours are observed during the summer, suggesting that this season may be most suitable for scheduling generator maintenance and downtime.



Figure 5.5: Yearly frequency of the top 10 peak net load hours for every country of the geographical scope. Every year of the considered time span results in at least one of the top 10 peak net load events when considering the entire scope of the research, while every country presents at least 4 different years with top peak net demand hours. This result underlies the importance of considering inter-annual variability when doing power system planning. Specific years appear to be more prevalent compared to others: 1985 and 1991 are present in every country, followed by 1989.

5.2. Analysis of LDES Dispatch Modeling

As described in Section 3.1.2, it is common for the storage to be depleted by the end of the optimization period. This approach is cost-effective, however, this leads to an unrealistic situation where no stored energy is carried over into the next optimization period, leaving the system unprepared for future demand. In reality, storage would need to be managed more conservatively to ensure energy availability across consecutive periods. This section compares different ways of modeling LDES dispatch and demonstrates that varying LDES dispatch model structures and initial assumptions can significantly impact results. The modeling options that are compared are presented in Section 4.3.3. EENS is calculated for each of these structures as explained in Section 4.3.4, while forced outages are implemented for nuclear and gas CCS units as described in Section 4.3.2.

To analyze storage operation, the charge levels of the storage were compared across various model structures for an LDES discharge duration of 100 hours. First, to prove the need for a strategy to regulate LDES dispatch, Figure 5.6 shows the charge and discharge cycles of hydrogen storage for the Netherlands where an E2P ratio of 100 hours is assigned to hydrogen. When no value is given to the stored energy, hydrogen cycles more frequently and is depleted by the end of the optimization horizon. In this case, hydrogen undergoes 10 full cycles throughout the year, meaning it is fully charged and depleted 10 times. As anticipated, the hydrogen reserve is exhausted early in the simulation, and despite recharging, the stored energy is not retained for potential energy shortages. The SoC remains below 0.35 and never reaches 1, as shown by the shallow cycles in Figure 5.6. This limitation may stem from the lack of an extended optimization horizon or incentives to encourage LDES charging, restricting charging to peak wind power periods. Consequently, this approach leads to under-utilization of the storage capacity, which poses risks during peak net demand periods when storage and reserves are essential. The calculated EENS value is 0.22 TWh, highlighting how this short-sighted dispatch fails to utilize the storage effectively.

Building on the previous paragraph, hydrogen must be dispatched in a way that ensures energy reserves are available during high-risk periods, conserving energy when it's not needed and replenishing reserves throughout the year. This section evaluates different approaches implemented during this research to determine the best method for meeting these requirements. The figures reveal that while both the 72-hour and 200-hour look-ahead periods capture some elements of storage system operation, neither performs well across the entire year. In particular, the 72-hour window severely underutilizes the storage system operation during large demand swings but still does not fully utilize the available capacity. These findings are also reflected in the calculated EENS values, which show a significant decrease when moving from the 72-hour to the 200-hour look-ahead dispatch. However, the EENS values remain similar between the 200-hour look-ahead dispatch and the dispatch with discharge/charge costs. The number of full hydrogen usage cycles aligns with these findings, as summarized in Table 5.1. In general, shortening the look-ahead period reduces the number of full storage cycles due to shallow cycling and underutilization. This implies that the look-ahead period should exceed the anticipated operational cycle of the storage system. However, for LDES, longer look-ahead periods are impractical because they significantly increase computational time and may not reflect realistic operational scenarios. The final model, which discourages discharging unless an energy shortage is imminent and prioritizes hydrogen charging after battery charging, better approximates the ideal operation of LDES. It also has the added benefit of reducing computational time.



Figure 5.6: The results show that without assigning value to stored energy, hydrogen storage undergoes 10 full cycles annually but is depleted by the end of the period. The SoC remains below 0.35 throughout, indicating under-utilization of storage capacity. This is due to limited charging opportunities, which may lead to insufficient storage during high demand. The calculated Expected Energy Not Supplied (EENS) value is 0.22 TWh, reflecting ineffective storage utilization under the current dispatch strategy.



Figure 5.7: Comparison of LDES dispatch with different look-ahead windows: a) 72-hour and b) 200-hour periods, alongside a model incorporating both charging and discharging costs. Both scenarios underutilize storage, with the 72-hour window using only 50% of capacity and the 200-hour window better handling large demand swings but still falling short of full utilization.

Table 5.1: Summary of EENS values and NFC for various LDES dispatch strategies. A higher number of NFC is calculated when no dispatch strategy is implemented for LDES. This is because LDES is depleted often as there is no incentive to keep the stored energy for potential risk events. However, the absence of such an incentive leads to a higher EENS. Extending the look-ahead window beyond the expected discharge duration significantly reduces EENS. However, the most effective approach to approximate optimal LDES dispatch is to implement both charging and discharging costs for LDES, leading to lower EENS and full utilization of storage capacity.

Method	NFC	EENS [GWh]
No method	10	220
Discharge cost - 72-hour look ahead window	2	196
Discharge cost - 200-hour look ahead window	4	48 ±2
Discharge cost - Charge cost	5	35 ±2

Incorporating both discharge and charging costs resulted in a more optimal dispatch for the LDES system. This strategy is then maintained throughout the rest of the analysis. With this approach, alongside the application of FO patterns, hourly simulations were conducted to determine the least-cost operation of the interconnected power system for the year 2050. The charging of energy storage and the transmission exports are both included in the electricity loads. The system dispatch results shown in Figure 5.8 highlight the temporal patterns of generation, storage, and load. The net load, calculated as the difference between load and VRE output, displayed variations over multiple days, weeks, months, and seasons (Figure 5.9). This reveals that most periods of the year experience insufficiencies in VRE resources, causing the combined solar, wind, and hydropower generation to fall short of meeting demand. Consequently, to bridge this gap, substantial amounts of energy are released by flexible resources such as biogas and gas, which act as a baseload, and storage. The daily net load frequently fluctuates, with a mean of 6.91 GWh/day and a standard deviation of 4.69 GWh/day. The daily net load is lowest in summer and peaks in winter and spring, with significant fluctuations spanning weeks or months.



Figure 5.8: Annual national dispatch results under Scenario 1. The black solid line represents The Netherlands' national load.



Figure 5.9: Net electricity demand in The Netherlands per day in 2050. The blue and red lines in the left figure represent 7-day and 30-day moving averages, respectively. For most of the year, VRE sources are insufficient to meet demand. To compensate, flexible resources such as biogas, gas, and storage systems release significant amounts of energy, often acting as a baseload. The daily net load varies considerably, with an average of 6.91 GWh/day and a standard deviation of 4.69 GWh/day.

5.3. Impact of LDES Discharge Duration on System's Adequacy

Section 5.1 presented the results from the capacity expansion model, which provided insights into the installed capacities of both battery and hydrogen storage systems across the countries within the study's geographical scope. As previously mentioned, the E2P ratio for batteries was constrained to less than 10 hours and similarly, the E2P ratio for hydrogen storage was also constrained to under 10 hours. In Section 5.2, the optimal method for dispatching energy stored in LDES technologies, such as hydrogen, was evaluated. This section examines the impact of varying discharge durations on the reliability of the power system, using this type of dispatch. The system operation has been simulated for different values of storage duration and the system reliability has been assessed by calculating the yearly EENS for each case. This analysis focuses on the Netherlands but could have been conducted for another country that belongs to the region considered.

First, since the beginning of the simulations, it was noted that the coefficient of variation α reached

a value of 0.3 with a small number of simulations, around 40 for almost every case. This was most likely since only one weather year was considered, reducing the uncertainty of the system. The uncertainty factor in these simulations was the FOs of gas and nuclear generators. However, the generation expansion resulted in the absence of nuclear power in The Netherlands and gas power plants contributing only 2% to the entire generation. This explains the minor variability in terms of loss of load time series between one simulation and the other. This is validated by comparing this result with the loss of load of other countries that are more heavily reliant on thermal generators, such as Germany. Gas power plants represent 10% of the German generation mix. Figure 5.10 shows the load loss taking place in Germany during the same year and for two different simulations. Load is curtailed at different times of the year and with a different magnitude, reflecting two different FO patterns for the two simulations. Big load curtailment can be seen for specific periods of the year, underscoring the presence of adverse weather conditions. This suggests that weather variability will be one of the major factors of resource adequacy failures in the future [94].



Figure 5.10: Loss of Load in Germany - Two simulations. Different colors represent different load-curtailment tranches. Comparing the two simulations, the timing and magnitude of the curtailment vary, reflecting distinct FO patterns in the simulations. However, significant load curtailment occurs during specific periods, highlighting the impact of adverse weather conditions.

Figures 5.12, 5.13, and 5.14 show the charging and discharging cycles of LDES, electricity demand, and production from various generation technologies over three weeks with multiple low wind power output periods (Figure 5.11). The Energy-to-Power (E2P) ratio of LDES is set at 50, 100, and 200 hours, respectively. From the 12th of October to the 22nd of October, and again from the 25th of October to the 1st of November, the wind is low, with significant drops on October 17th, 21st, 27th, and 30th. The phase from October 25th to November 1st results in a particularly alarming situation of the power system's reliability, as the wind capacity factor stays below 0.4 for 6 consecutive days. These days, generation sources such as biogas and gas CCUS contribute to meeting demand, supported by battery and hydrogen storage. It is noteworthy that transmission imports alone cannot provide sufficient electricity to prevent load curtailment. During periods of load shedding, the transmission network's capacity remains significantly underutilized, operating at only a tenth of its maximum capacity. This indicates that the entire power system is strained and countries are unable to support each other. Therefore, increasing transmission capacity would not mitigate load curtailment, as the overall system is under pressure.

LDES can provide power for several hours without recharging during these low-wind periods. This capability is crucial, especially during extended periods of low availability of renewable energy resources or generation capacity. Hydrogen storage typically depletes during peak demand and dunkelflaute, recharging when wind power returns. As the E2P increases, this type of storage offers firm capacity for supply reliability, assuming adequate storage capacity is available. As illustrated in the figures, extending the LDES E2P ratio helps solve some load curtailment events. Load curtailment time series, shown in the following pictures were analyzed to obtain the value of EENS and LOLE for each E2P case. In fact, while ENS provides a quantitative measure of the total energy shortfall, it does not capture the temporal characteristics of these shortages. Short-duration outages may indicate the need for flexible, short-term resources. Table 5.2 presents these calculated. It is worth mentioning that while EENS tends to plateau after 100 hours of discharge duration, LOLE continues to decline significantly. Moreover, increasing the discharge duration of LDES results in load curtailment events that last for fewer hours, as shown in the figures presented in Appendix A.1. Modeling a strictly increasing marginal cost for load shedding allows for a precise determination of both EENS and LOLE [95]. In this context, energy-limited resources are optimized to minimize LOLE [95]. The reduction in LOLE with higher E2P ratios for LDES can be attributed to the contribution of hydrogen in resolving some load curtailment events. As more energy is made available to the grid, this improvement is also reflected in the NFC values for each E2P scenario, as shown in Table 5.2. With a hydrogen capacity of 3.5 GW, as depicted in Figure 5.2, the NFC values correspond to discharge energies of 1400 GWh, 1750 GWh, and 2100 GWh for E2P ratios of 50, 100, and 200 hours, respectively.



Figure 5.11: Wind power output during the 3 weeks in NLD. The values are normalized to the installed power capacity. In this period, the power output of offshore and onshore wind remains below 40% of the total generation for ten days (12th Oct. - 22nd Oct.). After a few days of increased power output (never higher than 60% of the total production), the capacity factor lowers again for 5 days (25th Oct. - 1st Nov.)



Figure 5.12: The figure shows the generation and imports (top) alongside the exports and electricity demand (bottom) for the Netherlands from October 9, 1984, to November 1, 1984. The E2P ratio is set at 50 hours. The black lines illustrate the discharge and charging power of hydrogen. During this period, load curtailment events show more pronounced peaks and a wider distribution. Notably, there are no export activities during these times.



Figure 5.13: The figure shows the generation and imports (top) alongside the exports and electricity demand (bottom) for the Netherlands from October 9, 1984, to November 1, 1984. The E2P ratio is set at 100 hours. The black lines illustrate the discharge and charging power of hydrogen. In this case, load curtailment events happening on the 28th of October are solver, leading to a lower LOLE.



Figure 5.14: The figure shows the generation and imports (top) alongside the exports and electricity demand (bottom) for the Netherlands from October 9, 1984, to November 1, 1984. The E2P ratio is set at 200 hours. The black lines illustrate the discharge and charging power of hydrogen. In this case, almost all load curtailment events are solved, leading to a lower LOLE and allowing for exports.

Table 5.2: Calculated values of EENS and LOLE for different E2P ratios. Increasing the discharge duration of LDES from 50 to 100 hours reduces EENS considerably. The LOLE is also reduced as the dispatch of hydrogen solves some load curtailment events.

E2P [h]	NFC	EENS [GWh]	LOLE [h]
50	8	82 ±4	155 ±3
100	5	36 ±2	118 ±5
200	3	30 ±2	86 ±6

5.4. Varying power capacity, comparison with SDES

As seen in Section 2.3.2, when wind and solar power share reaches 90 % penetration in the energy mix, large, long-duration storage will be needed to meet electricity demand fully. However, the energy transition will probably go through different phases, where different technologies and flexibility options are employed. These phases, characterized by the share of low-carbon power generation, influence the required storage discharge duration [96]. In the first phase, a reduction in conventional plants— previously responsible for system regulation through generator inertia—occurs, shifting the focus to energy storage systems with durations of less than one hour. Many European countries, including Germany, are currently in this phase. For example, by 2020, more than 50% of frequency regulation in Germany was provided by batteries [97]. During the second phase, the levelized cost of electricity from renewables plus storage drops below that of fossil fuels, leading to widespread deployment of storage. Here, alternative technologies may play a stronger role due to the discharge requirements of the system. This motivates the examination of reliability metrics while varying the power and energy capacity of LDES and SDES. This section provides a resource adequacy assessment, comparing SDES and LDES based on reliability metrics such as EENS and LOLE. The analysis examines how varying the power and energy capacities of both storage types impacts these metrics.

To evaluate the effect of installed power capacity on supply security, the system's hydrogen capacity was halved, reallocating the remaining capacity to batteries. This change also altered the installed storage, linked to the installed power capacity by the E2P ratio, which is 10 hours for the case of the battery, and 200 hours for the case of hydrogen with a 200-hour discharge duration. The simulations yielded an EENS of 11 \pm 1 GWh and a LOLE of 100 \pm 2 hours. Notably, although the EENS value is lower for the E2P = 200 hours case (see Table 5.2), the LOLE value is higher. This is evident from Figure 5.15, which illustrates the magnitude and frequency of load curtailment duration for this type of dispatch. Compared to Figure A.3, load shedding peaks have been reduced, but some curtailment has shifted over time, increasing LOLE. Longer curtailment events, such as those lasting 18 or 19 hours, occur more frequently in this scenario, compared to 17-hour events in the E2P = 200-hour case. Thus, increasing installed power capacity reduces EENS but has less impact on LOLE.

To investigate whether this EENS reduction is due to increased battery power rather than storage, another scenario reduced LDES installed energy to one-fourth of its original value, with the remainder allocated to batteries. This resulted in an EENS value of 0.45 \pm 0.01 GWh and a LOLE of 42 \pm 1 hours. Figure 5.16 shows the total load shedding and its frequency distribution for this particular case. Increasing the energy storage capacity results in solving most of the load curtailment events and in reducing peak events.

By looking at these results, an immediate conclusion that could be drawn is that LDES is not necessary, as increasing the amount of energy capacity stored in batteries contributes to considerably decreasing the EENS. To understand better this point, a final test was conducted, where LDES storage capacity was halved to its original value, with the remaining part given to batteries. The simulations for this case resulted in an even smaller EENS of 0.25 ±0.1 GWh and a LOLE of 25 ±13 hours. The longest curtailment event was only 9 hours, compared to 18 hours in the previous scenario. This suggests that while battery capacity improves short-term reliability, LDES remains critical for ensuring supply security during extended periods of low renewable generation. Battery resources complement LDES by providing short- and mid-term flexibility, ensuring that LDES is only dispatched during periods when the limited discharge duration of batteries is insufficient. The reduced LOLE in cases with higher LDES capacity indicates that for prolonged low-wind conditions, batteries alone may lead to load curtailment, while LDES helps prevent shortages by providing electricity during sustained periods of low renewable output.



Figure 5.15: Load Shedding Distribution - Case of halved LDES power capacity. The figure illustrates the impact of halving the hydrogen power capacity and increasing battery capacity on load curtailment events. The shaved peaks and shifted load curtailment over time are visible, demonstrating the effect on EENS and LOLE. Increasing the power capacity of SDES technologies can be an effective way to decrease EENS, but not LOLE.



Figure 5.16: Load Shedding Distribution - Case of LDES energy capacity reduced to 1/4th. The figure shows a significant reduction in peak load curtailment, highlighting the role of battery energy capacity in reducing EENS and addressing shorter-duration curtailment events.



Figure 5.17: Load Shedding Distribution - Case of halved LDES energy capacity. The figure shows the reduction in EENS and LOLE reinforced by the complementary role of SDES and LDES. In particular, the participation of LDES seems to be related to a reduction in LOLE.

6

Discussion of the Results

In the previous chapter, the results obtained from the various simulations were presented. This chapter discusses how the finding can answer the research question: *What are the capacity value drivers of long-duration electricity storage for an interconnected European power system in 2050 in a context of high penetration of renewables and future climate?*. Section 6.1 will first analyze the background and scope of the project to provide context for the findings. Following this, Section 6.2 will focus on the main outcomes of the model and simulations. Section 6.3 will address the socio-economic implications of these results, exploring how the deployment of storage technologies could affect power system costs and energy security. Finally, the limitations of the research and potential areas for improvement will be discussed in Section 6.4.

6.1. Project background and scope

The problem that this thesis tries to solve is to assess to what extent long-duration electricity storage can contribute to the resource adequacy of a European interconnected system in 2050, in a context of uncertain and future climate. As climate concerns among governments are driving energy policies toward more sustainable electricity production with significant CO2 reductions, these long-term goals are being developed, tested and studied using advanced energy modeling tools, which predict a global increase in renewable energy production. This includes hydropower, biomass, geothermal, but most importantly, a substantial expansion of wind and solar power generation. However, solar and wind energy, known as variable renewable energy sources, present challenges for power sector management due to their variability across different timescales and their non-dispatchable nature, meaning operators cannot control their output beyond switching them off or reducing their output voluntarily. VRES significantly impact the power sector because they are prioritized in the merit order due to their zero marginal cost, necessitating that the remaining load is met by other dispatchable technologies.

The key challenge for power systems is to maintain reliability continuously through system planning and operation. Resource adequacy deals with the long-term planning of the power systems, as it involves the ability to meet long-term demand while considering supply-demand uncertainties. Due to the increasing share of VRES in the generation mix and the phase-out of traditional thermal generators, the weather will increasingly become a major factor in resource adequacy failures. Consequently, it is indispensable for energy planners to take into account meteorological uncertainty when doing resource adequacy assessments and recognizing the primary meteorological factors contributing to power system strain. However, future climate models present variability from one to the other and can represent a new source of uncertainty. For this reason, calibrating and adjusting historical climate data and scenarios to account for the likelihood calibrating and adjusting historical climate data and scenarios to account for the likelihood of climate change impacts can give useful information in assessing how past extremes can affect the power grid in the future. Moreover, to cope with an increasing reliance on weather-dependent renewables, flexibility options such as electricity storage, interconnection, and demand response will be needed to guarantee adequate future power systems. LDES is becoming increasingly important for ensuring grid adequacy, especially as the penetration of variable renewable energy sources grows, necessitating different storage technologies for various discharge durations to optimize electricity costs. There is an ongoing debate around how much longduration electricity storage is needed and what form it should take, particularly in comparison to other flexibility options like short-term storage or firm low-carbon energy sources, such as nuclear power. No prior study has thoroughly examined the benefits of LDES and the key factors driving its advantages in a future interconnected European power system. This research aimed to fill that gap by conducting energy assessments for an interconnected European grid in 2050, incorporating LDES under various scenarios. To achieve a more precise representation of future conditions, the analysis factored in weather variability and the effects of anthropogenic climate change.

6.2. Model and Simulations Findings

This section presents the results of the research, which was conducted in two main parts: first, the development of a capacity expansion model to determine the most cost-effective mix of generation, storage, and transmission investments; and second, a resource adequacy assessment of this system under various scenarios.

Capacity Expansion Results

The capacity expansion model simulated investments in generation and transmission capacity for a European interconnected system, based on assumptions about future electricity demand, technology costs, performance, and generator availability. This model was developed to endogenously determine the future levels of short-duration and long-duration energy storage (SDES and LDES), modeled as batteries and hydrogen respectively, that will be necessary for the future power system. Existing projections, such as those from the Ten-Year Network Development Plan, often focus on hydrogen needs for the entire energy system, not just the power sector. The model results show that as the share of variable and inflexible low-carbon electricity sources—such as solar, wind, and hydropower—exceeds 90%, conventional thermal generators are significantly reduced. In fact, VRE sources, such as wind and solar, fully utilize their maximum allowed capacity in the model's capacity expansion constraint, while other technologies like nuclear and gas power did not reach their full potential in all countries. Lithium-ion batteries, with the lowest installed capacity cost, made up the largest share at 61 GW, while hydrogen storage contributed 30 GW.

The expansion results were based on 10 years of weather data (from 1985 to 1995). Analysis of the top 10 netload events across different years revealed that each year contributed to at least one peak net load event, with each country experiencing peak demand hours in at least four distinct years. This highlights the importance of using multiple years of weather data when planning capacity expansions to ensure the system is resilient under varying conditions. Additionally, a seasonal analysis showed that the majority of peak net load hours occurred in winter and autumn. This provides valuable insight for energy planners, suggesting that long-duration energy storage systems should be strategically charged during the summer to be ready for discharge during the high-demand periods in winter and autumn.

Impact of LDES dispatch on resource adequacy

In Section 5.2, different approaches to modeling LDES dispatch were analyzed to highlight how varying model structures and initial assumptions can significantly affect the results. LDES must be dispatched in a way that ensures energy reserves are available during critical periods, conserving energy when demand is low and replenishing reserves throughout the year. Without a specific dispatch strategy, the typical economic dispatch tends to discharge LDES aggressively while delaying recharging, making it vulnerable to short-term outages and future system requirements. While this may be suitable for short-duration storage, where cost savings come with a slight reduction in reliability, it is unrealistic for systems relying on LDES or seasonal storage, which require longer planning horizons.

Three dispatch strategies for LDES were compared. In all cases, LDES was the last resource to be discharged, only used before resorting to load shedding. The first two strategies featured lookahead periods of 72 hours and 200 hours, respectively, but neither had a dedicated charging strategy. The third strategy also used a 72-hour look-ahead period but introduced a charging incentive for both SDES and LDES, prioritizing SDES for charging. The results indicated that proactive charging during lower-demand periods, such as the shoulder seasons, is important to ensure enough energy reserves for high-demand periods. Shorter look-ahead periods led to fewer full storage cycles due to shallow cycling and underutilization. This conclusion is supported by the fact that the 200-hour look-ahead strategy performed quite well, resulting in an EENS value only slightly larger than that of the third strategy, which included proactive charging. However, extending the look-ahead period too far becomes impractical, especially when modeling LDES or seasonal storage, which is likely to require operational procedures akin to those used in hydrothermal systems. [98].

Varying Discharge Duration and Comparison with SDES

In Section 5.4, resource adequacy assessments were performed to explore how variations in installed power and energy capacity of LDES and SDES impact reliability metrics such as EENS and LOLE.

The results demonstrate that increasing energy capacity is more effective at reducing EENS than adjusting installed power alone. Additionally, extending the discharge duration of LDES significantly mitigates long-duration load curtailment events. For instance, increasing the hydrogen discharge duration from 50 to 200 hours reduced EENS by 34% and LOLE by 42 %. This suggests that storage capacity value is closely tied to both power rating and energy storage capacity to adequately cover peak demand periods. Table 6.1 presents the frequency distribution of load shedding durations across different time ranges for all the cases analyzed in this study, highlighting a clear pattern: as the LDES discharge duration increases, LOLE decreases, and load curtailment events become shorter.

However, when comparing the EENS reduction achieved by increasing the E2P ratio of LDES with the reduction obtained by enhancing the energy capacity of SDES, as shown in Section 5.4, one notable observation emerges: increasing the E2P ratio of LDES does not reduce the peak magnitude of the main load curtailment event. Increasing the energy capacity of SDES could contribute to reducing the magnitude of these events and their duration, however, it may not be economically viable, as shown in Section 2.3.1. Furthermore, comparing the halved power case to the E2P = 200 hours case reveals a 63% reduction in EENS for the E2P = 200 hours case, though accompanied by a 10% increase in LOLE. This indicates that increasing the power capacity of SDES technologies is effective at mitigating the peaks of load curtailment events, thereby lowering EENS while increasing the discharge duration of LDES has more impact on the LOLE rather than the EENS.

To summarize, storage systems incorporating both battery and hydrogen storage offer greater reliability than one relying on a single energy storage solution. This approach leverages the strengths of each type: batteries provide quick response and are ideal for short-term energy needs, while hydrogen can cover longer durations. This diversification reduces the risk associated with relying on a single storage technology, particularly during periods of low renewable energy availability or high demand. However, power systems must optimally balance SDES and LDES to ensure adequacy during extreme weather events. This balance is challenging as the benefits of adding more SDES or more LDES have to be compared with the disadvantages. For example, increased SDES can lead to an increased electricity demand due to the electricity required to charge the batteries and potential grid congestion, while LDES can contribute to better VRES integration. Ultimately, these findings suggest that increasing the energy capacity of SDES has a more significant impact on reducing EENS, whereas increasing the discharge duration of LDES is crucial for achieving the greatest reduction in LOLE. This raises questions about which metric—EENS or LOLE—should be prioritized when evaluating power system adequacy. Table 6.1: The frequency for certain ranges (in hours) of load shedding duration is presented. It can be noticed that increasing the discharge duration of LDES can effectively reduce the duration of load-shedding events while increasing the power capacity of SDES contributes to a lower EENS but a higher chance of having longer load curtailment events. A solution is either increasing the energy capacity of SDES, which may not be economically viable or increasing the discharge duration of LDES.

CASE	1-2	3-5	6-8	9-11	12-15	16-19
E2P = 50 hours	2	4	1	4	0	5
E2P = 100 hours	2	3	2	4	0	3
E2P = 200 hours	2	5	3	3	0	1
Halved Power Capacity	1	5	1	1	0	3
1/4th Energy Capacity	0	1	0	1	1	1
Halved Energy Capacity	1	2	2	1	0	0

6.3. Socio-Economic Impact

The findings discussed earlier in this paper raise important questions regarding which metric, EENS or LOLE, should take precedence when evaluating system adequacy and defining a reliability standard. As a matter of fact, according to Article 25 of Regulation 2019/943, Member States are required to have a reliability standard in place when implementing capacity mechanisms [99]. The purpose of this standard is to determine the capacity needed within the capacity mechanisms. These mechanisms have become a key regulatory tool in liberalized power systems undergoing decarbonization, ensuring that an adequate resource mix is available to meet the reliability targets set by regulators [99]. The standard is calculated using metrics like the Value of Lost Load (VOLL) and the Cost of New Entry (CONE) over a defined period and is expressed in terms of EENS and LOLE, which indicate the optimal level of security of supply [99]. This procedure is depicted in Figure 6.1, which illustrates how the optimal capacity volume is determined by maximizing the net social benefit of electricity. Social benefit is maximized when the marginal cost of capacity equals the marginal benefit. The marginal cost is primarily driven by the fixed costs of adding an MW of peaking capacity (CONE), while the marginal benefit reflects the value of outages avoided through additional capacity, quantified as VOLL multiplied by LOLE.

This research highlighted that today's reliability events are more diverse in size, frequency, duration, and timing. Understanding these variations is essential for selecting appropriate resource solutions. Mitigation strategies need to be tailored to address these different scenarios, as each resource type brings distinct capabilities, as seen when comparing LDES and SDES. For these reasons, grid planners and regulators must have a clear understanding of the costs associated with achieving different reliability targets, recognizing that each resource has specific strengths and limitations, which need to be optimally balanced for modern power system adequacy. In the future, both EENS and LOLE might need to be considered together for a comprehensive approach to modern power system adequacy.



Figure 6.1: Economic equilibrium determining the reliability standard

The simulations revealed that the dispatch of LDES significantly impacts EENS, raising important economic considerations. To optimize dispatch, LDES should operate on an annual cycle, with the hydrogen system completing about 5 cycles per year. However, the uncertain cost competitiveness of LDES raises concerns about its widespread deployment. As storage duration increases and dispatch becomes more optimized, fewer operating cycles of LDES are expected each year, as discussed in Section 5.2. This necessitates finding incentives for storage operators to maintain adequate storage levels, even when it is used only a few times annually. Additionally, one of the main barriers to energy storage investment is the gap in EU legislation, which results in grid fees being applied both during charging and discharging [100]. This issue of double grid fees is a significant concern for investors in countries where both generation and consumption are taxed [100]. Moreover, the 2019 E-Directive prohibits storage ownership and operation to network operators, such as TSOs and DSOs, from owning storage facilities [100]. As a result, storage plant owners must compete in electricity markets, often with less access to critical information than SOs, such as wind and load forecasts or network and plant statuses [100].

To promote the construction of energy storage facilities, capacity payments or other non-market incentives are also essential to support storage investments. These incentives are particularly important because the societal benefits of energy storage, such as reduced production costs, lower GHG emissions, and grid stabilization, do not always generate market revenues for storage owners [101]. To bridge the current cost gap and address technological uncertainties in the emerging LDES market, governments and industry leaders must stimulate market development by fostering an investment-friendly ecosystem. Three critical areas for support are long-term system planning to generate investment signals, scaling up deployments to kick-start the market, and developing mechanisms to capture and monetize the value of LDES.

6.4. Limitations

It is essential to acknowledge the limitations of this research to better understand how the results might translate to real-world scenarios. Below, several of these limitations are outlined, along with potential areas for improvement.

The planning mode results were obtained using heuristic clustering of timesteps, chosen to reduce computational time. A 6-hour time resolution was used for most of the ten years of weather data considered. However, this may have introduced an end effect when modeling photovoltaics, as solar capacity factors can vary on shorter timescales. This lower resolution might lead to an overestimation of storage requirements. Exploring alternative time clustering techniques and comparing the results could offer insights into the system's actual storage needs.

The weather data used in this model was based on historical data calibrated with future climate projections, which carry inherent uncertainty. Incorporating multiple climate models to compare outcomes could enhance the robustness of the results. Additionally, the hydropower dataset relied on historical records due to the absence of future projections. Given the potential for climate change to cause droughts, integrating hydropower timeseries that account for this phenomenon would provide a more realistic evaluation of future power system stress.

A degree of uncertainty was introduced through the implementation of forced outage patterns. However, the results showed that resource adequacy was less sensitive to this factor, primarily due to the system's growing dependence on weather patterns. To test system reliability further, adding more uncertainty variables could be beneficial. For example, using more than one year of weather data for resource adequacy assessments, or introducing uncertainty into demand timeseries, could improve the analysis.

The dispatch strategy for LDES could also be optimized. Although the current approach performs well in managing reserves and charging when needed, it does not accurately reflect real-world electricity market dynamics. For instance, the current model assumes a negative charging cost, implying that storage operators earn revenue when charging, which isn't realistic. A more sophisticated approach could be the one discussed in [73], which involves introducing a variable for the value of stored energy. In this case, if the marginal price of energy falls below the value assigned to stored energy, the storage device would charge, while if the marginal price exceeds this value, the device would discharge.

Finally, while the assumptions and inputs for this study were carefully defined and drawn from credible sources, the long-term horizon to 2050 introduces considerable uncertainty. Therefore, the findings should be interpreted with caution, especially when considering long-term predictions.

Conclusion and Recommendations

The problem that this thesis tries to solve is to assess to what extent long-duration electricity storage can contribute to the resource adequacy of a European interconnected system in 2050, in a context of uncertain and future climate. LDES is becoming increasingly important for ensuring grid adequacy, especially as the penetration of variable renewable energy sources grows, necessitating different storage technologies for various discharge durations to optimize electricity costs. There is an ongoing debate around how much long-duration electricity storage is needed and what form it should take, particularly in comparison to other flexibility options like short-term storage or firm low-carbon energy sources, such as nuclear power. No prior study has thoroughly examined the benefits of LDES and the key factors driving its advantages in a future interconnected European power system. This research aimed to fill that gap by conducting energy assessments for an interconnected European grid in 2050, incorporating LDES under various scenarios. This objective is described by the research question: What are the capacity value drivers of long-duration electricity storage for an interconnected European power system in 2050 in a context of high penetration of renewables and future climate?. To achieve a more precise representation of future conditions, the analysis factored in weather variability and the effects of anthropogenic climate change. The goal was to obtain a deeper understanding on what can influence the ability of LDES to bring resource adequacy to the system. This chapter will provide the final conclusions of this project.

7.1. Conclusion

The core challenge for power systems is ensuring continuous reliability through careful planning and operation. Resource adequacy focuses on long-term planning, addressing the system's capacity to meet future demand while accounting for supply-demand uncertainties and it consists of transmission adequacy and generation adequacy. In contrast, system security handles the short-term reliability of the power supply. This research places particular focused on generation adequacy. Various metrics, including Expected Energy Not Supplied (EENS), Loss of Load Expectation (LOLE), and Loss of Load Probability (LOLP), can be calculated and compared against European benchmark values. This study quantified EENS and LOLE for a power system comprising eight European countries, with electricity demand projections adjusted to reflect those forecasted for 2050. The decision to focus on EENS and LOLE was driven by the fact that they capture two key dimensions of load curtailment: severity and duration.

A capacity expansion model was developed using the energy modeling framework Calliope. This model incorporated ten years of weather data, employing calibrated historical data and climate projections to assess how past extreme weather events might influence future grid performance, ultimately helping to design a more resilient system. The capacity expansion model simulated investments in generation and transmission capacity across the interconnected system and endogenously determined the future levels of both short-duration and long-duration energy storage (SDES and LDES). the study analyzed seasonal patterns in peak net load, defined as total demand minus total variable renewable energy (VRES) generation. The results showed that the highest risks to system stability occurred during peak

demand periods in autumn and winter, suggesting that summer could be an optimal time to schedule generator maintenance and charge LDES. The study analyzed seasonal patterns in peak net load, defined as total demand minus total variable renewable energy (VRES) generation. The results showed that the highest risks to system stability occurred during peak demand periods in autumn and winter, suggesting that summer could be an optimal time to schedule generator maintenance and charge LDES.

After establishing the installed capacity for VRES, thermal generators, and storage, a Monte Carlo resource adequacy assessment was performed. The chosen period, spanning from June 1, 1984, to June 1, 1985, included a particularly severe winter in 1985, marked by extreme weather that heavily impacted energy supply and demand. This scenario provided insights into the role of long-duration storage during system stress. Forced outage patterns for thermal generators were incorporated, however, the resource adequacy results showed a low sensitivity to these patterns, due to the large dependence of the grid on VRES. Multiple tranches of load curtailment were modeled with increasing marginal costs. The capacity for each tranche was distributed proportionally across countries, creating a strategy that minimized load shedding and maintained a unique LOLE value. An analysis of imports and exports during periods of load curtailment revealed that transmission capacity was never fully utilized during these events. This suggests that increasing transmission capacity alone would not resolve the issue of load curtailment.

By simulating different LDES dispatch strategies, it became evident that the operational profile of LDES plays a critical role in the system's resource adequacy. The operational profile of LDES is characterized by fewer cycles, strategic dispatch aimed at minimizing EENS, and a strong dependence on the value attributed to stored energy. Optimal dispatch strategies aim to limit discharging to periods of potential energy shortages while encouraging strategic charging. LDES technologies, such as hydrogen storage, tend to cycle more frequently when the stored energy lacks a defined monetary value, often resulting in shallower cycles and underutilization. Such frequent cycling can deplete storage reserves prematurely, leading to suboptimal performance during high-demand periods. The results also showed that extending the look-ahead period beyond the expected storage duration could improve performance, though this approach is often impractical for LDES. In scenarios with controlled charging and discharging, LDES was dispatched more conservatively, functioning as a buffer during peak demand and cycling less frequently. This resulted in enhanced supply security and lower EENS. Additionally, the number of cycles that LDES undergoes annually is influenced by the energy-to-power (E2P) ratio. As the E2P ratio increases, the frequency of cycles decreases. Consequently, policy and market design play a crucial role in shaping the operational profile of LDES. Current policies supporting LDES need to be strengthened, and market mechanisms should be designed to provide adequate financial returns to encourage investment.

Varying the E2P ratio of LDES allowed to understand that the duration of energy storage significantly affects both EENS and LOLE. A higher E2P ratio and increased storage capacity are key to effectively reducing EENS. While power capacity is typically the focus in resource adequacy assessments, the value of storage is closely tied to both its power rating and energy storage capacity to meet peak demand. Storage systems with higher E2P ratios are more capable of sustaining energy output during periods of low renewable generation or high demand, making them more effective at reducing EENS. This is especially critical during extended periods of low renewable energy output, such as prolonged cloudy or windless days. For example, increasing the storage duration from 50 to 200 hours reduced EENS by 34% and LOLE by 42%. However, beyond a certain E2P threshold, improvements in EENS tend to plateau.

The simulations also showed that LDES and SDES serve complementary roles in reducing EENS. SDES technologies are particularly effective in addressing short-term disruptions, such as fluctuations in wind power output. While increasing the E2P ratio of LDES does not significantly reduce the peak magnitude of the largest load curtailment events, increasing the power capacity of SDES can help mitigate the magnitude and duration of these events, though it may not always be economically viable due to the higher energy investment costs. The lower power investment costs of SDES could be exploited to lower EENS, however, LDES plays a more substantial role in reducing LOLE. This re-

search highlighted the increasing diversity of reliability events in terms of size, frequency, duration, and timing. Understanding these variations is essential for selecting appropriate resource solutions. Grid planners and regulators must carefully consider the costs associated with achieving different reliability targets, balancing the strengths and limitations of each resource type to ensure modern power system adequacy.

7.2. Answer to Research Question

When projecting 1985, which presented several episodes of cold snap and wind drought, into the future the resource adequacy assessment conducted showed that power systems with a high penetration of renewables need large amounts of storage. LDES significantly contributes to the resource adequacy of a power system by providing a reliable source of backup power. This ensures that there is sufficient generation capacity available to meet peak demand, even during periods of low renewable generation. In this context,LDES helps maintain adequate generation capacity to meet peak demand, thus displacing the need for firm generation capacity. This research demonstrated that the value drivers of storage depends not only on the power rating of the storage but also on the energy storage capacity and the discharge duration as it has to cover the duration of the peak demand. In addition to that LDES capacity value depends also on the way this technology is dispatched. Effective dispatch optimization minimizes EENS by ensuring that stored energy is utilized when it is most needed. The document shows that optimal dispatch strategies significantly reduce EENS, emphasizing the importance of smart energy management to maximize the value of LDES. However, while LDES can reduce LOLE by covering long-term energy shortages, it may not always mitigate the magnitude of peak load curtailment events.

Finally, the combination of both storage types leads to greater reductions in EENS compared to either type alone. The study showed that the best storage technologies to balance the system depend on the duration of deficit events it is designed to mitigate. Therefore, the integration of LDES with SDES enhances overall capacity value by providing a comprehensive storage solution that addresses both short-term and long-term energy needs. SDES can handle rapid, short-duration fluctuations, while LDES covers extended periods of energy deficit. This underlines the necessity of integrating various storage solutions.



Appendix

A.1. Impact of LDES Discharge Duration on RA - Load Sheddig Distribution



Figure A.1: Load shedding and load shedding frequency distribution for an E2P ratio of 50 hours.



Figure A.2: Load shedding and load shedding frequency distribution for an E2P ratio of 100 hours.



Figure A.3: Load shedding and load shedding frequency distribution for an E2P ratio of 200 hours.

Bibliography

- [1] Wetterzentrale. Historical pressure and temperature map of europe, 2024. URL https://www. wetterzentrale.de/de/default.php.
- [2] Hannah C. Bloomfield, David J. Brayshaw, Matthew Deakin, and David Greenwood. Hourly historical and near-future weather and climate variables for energy system modelling. *Earth System Science Data*, 14:2749–2766, 6 2022. ISSN 18663516. doi:10.5194/essd-14-2749-2022.
- [3] European Union Horizon 2020. Primavera, 2021. URL https://www.primavera-h2020.eu/.
- [4] European Network of Transmission System Operators for Electricity ENTSO-E. Tyndp 2022 scenario report, 2022. URL https://2022.entsos-tyndp-scenarios.eu/wp-content/uploads/2022/ 04/TYNDP2022_Joint_Scenario_Full-Report-April-2022.pdf.
- [5] United Nations Climate Change. Cop28 agreement signals "beginning of the end" of the fossil fuel era, 2023. URL https://unfccc.int/news/ cop28-agreement-signals-beginning-of-the-end-of-the-fossil-fuel-era.
- [6] European Network of Transmission System Operators for Electricity. Winter outlook 2023-2024, summer review 2023, 2023.
- [7] United Nations Sustainable Development. Goal 7: Affordable and clean energy, 2015. URL https://www.unep.org/topics/sustainable-development-goals/ why-do-sustainable-development-goals-matter/goal-7-affordable.
- [8] E. Ciapessoni, D. Cirio, A. Pitto, M. Van Harte, and M. Panteli. Power system resilience: definition, features and properties, 2023. URL https://cse.cigre.org/cse-n030/ power-system-resilience-definition-features-and-properties.html#:~:text=Reliability-,Adequacy, and%20in%20the%20amount%20desired.
- [9] Andrew Keane, Michael Milligan, Chris J Dent, Bernhard Hasche, Claudine D'Annunzio, Ken Dragoon, Hannele Holttinen, Nader Samaan, Lennart Soder, and Mark O'Malley. Capacity value of wind power. *IEEE Transactions on Power Systems*, 26(2):564–572, 2010.
- [10] Paul Albertus, Joseph S Manser, and Scott Litzelman. Long-duration electricity storage applications, economics, and technologies. *Joule*, 4(1):21–32, 2020.
- [11] European Commission. Energy and the green deal, 2022. URL https://commission.europa. eu/strategy-and-policy/priorities-2019-2024/european-green-deal/energy-and-green-deal_en# documents.
- [12] European Commission. Identification of appropriate generation and system adequacy standards for the internal electricity market final report, 2016. URL https://energy.ec.europa.eu/system/files/ 2016-07/Generation%2520adequacy%2520Final%2520Report_for%2520publication_0.pdf.
- [13] Mojtaba Shirvani, Ahmad Memaripour, Mostafa Abdollahi, and Asadollah Salimi. Calculation of generation system reliability index: Expected energy not served. *Life Sci. J*, 9(4):3443–3448, 2012.
- [14] European Network of Transmission System Operators for Electricity. Mid-term adequacy forecast 2018: Appendix 1: Methodology and detailed results - 2018 edition, 2018.
- [15] European Network of Transmission System Operators for Electricity. European resource adequacy assessment - 2023 edition - annex 2: Methodology, 2023. URL https://www.entsoe.eu/ outlooks/eraa/2023/report/ERAA_2023_Annex_2_Methodology.pdf.

- [16] Gord Stephen, Simon H. Tindemans, John Fazio, Chris Dent, Armando Figueroa Acevedo, Bagen Bagen, Alex Crawford, Andreas Klaube, Douglas Logan, and Daniel Burke. Clarifying the interpretation and use of the lole resource adequacy metric. In 2022 17th International Conference on Probabilistic Methods Applied to Power Systems (PMAPS), pages 1–4, 2022. doi:10.1109/PMAPS53380.2022.9810615.
- [17] Pentalateral Energy Support Group 2. Generation adequacy assessment, 2015. URL https://www.swissgrid.ch/dam/swissgrid/current/News/2015/PLEF GAA-report en.pdf.
- [18] Sebastian Gonzato, Kenneth Bruninx, and Erik Delarue. The effect of short term storage operation on resource adequacy. *Sustainable Energy, Grids and Networks*, 34, 6 2023. ISSN 23524677. doi:10.1016/j.segan.2023.101005.
- [19] Juan Pablo Carvallo, Nan Zhang, Benjamin D Leibowicz, Thomas Carr, Sunhee Baik, and Peter H Larsen. A guide for improved resource adequacy assessments in evolving power systems: Institutional and technical dimensions. 2023.
- [20] LLP Linklaters. Capacity mechanisms. reigniting europe's energy markets, 2014.
- [21] Elizabeth Potter. Examining The California And Texas Power Outages. PhD thesis, 2021.
- [22] Adam X Andresen, Liza C Kurtz, David M Hondula, Sara Meerow, and Melanie Gall. Understanding the social impacts of power outages in north america: a systematic review. *Environmental Research Letters*, 18(5):053004, 2023.
- [23] Michiel De Nooij, Rogier Lieshout, and Carl Koopmans. Optimal blackouts: Empirical results on reducing the social cost of electricity outages through efficient regional rationing. *Energy Economics*, 31(3):342–347, 2009.
- [24] Dave Jones Sarah, Brown. European electricity review, 2024. URL https://ember-climate.org/ insights/research/european-electricity-review-2024/.
- [25] California Energy Commission. A peek at net peak, 2021. URL https://www.energy.ca. gov/data-reports/energy-insights/peek-net-peak#:~:text=Peak%20Demand%3A%20The% 20highest%20amount,scale%20wind%20and%20solar%20generation.
- [26] Tiziano Gallo Cassarino, Ed Sharp, and Mark Barrett. The impact of social and weather drivers on the historical electricity demand in europe. *Applied energy*, 229:176–185, 2018.
- [27] Chris Dent. Managing climate uncertainty, 2024. URL https://www.nerc.com/comm/RSTC/ PAWG/PAF_Forum-2023-Day_3_Presentations.pdf.
- [28] Philip Heptonstall, Robert Gross, and Florian Steiner. The costs and impacts of intermittency– 2016 update. *London: UK Energy Research Centre*, 2017.
- [29] Maximilian Bernecker, legor Riepin, and Felix Müsgens. Modeling of extreme weather events towards resilient transmission expansion planning. In 2022 18th International Conference on the European Energy Market (EEM), pages 1–7. IEEE, 2022.
- [30] H. C. Bloomfield, C. C. Suitters, and D. R. Drew. Meteorological drivers of european power system stress. *Journal of Renewable Energy*, 2020:1–12, 8 2020. ISSN 2314-4386. doi:10.1155/2020/5481010.
- [31] Stefan Pfenninger. Dealing with multiple decades of hourly wind and pv time series in energy models: A comparison of methods to reduce time resolution and the planning implications of inter-annual variability. *Applied Energy*, 197:1–13, 2017. ISSN 03062619. doi:10.1016/j.apenergy.2017.03.051.
- [32] MO Molina, C Gutiérrez, M Ortega, and E Sánchez. Summer heatwaves, wind production and electricity demand in southern europe: climatic conditions and impacts. *Environmental Research Communications*, 5(8):085005, 2023.

- [33] Srihari Sundar, Michael T. Craig, Ashley E. Payne, David J. Brayshaw, and Flavio Lehner. Meteorological drivers of resource adequacy failures in current and high renewable western u.s. power systems. *Nature Communications*, 14, 12 2023. ISSN 20411723. doi:10.1038/s41467-023-41875-6.
- [34] Stefan Pfenninger and James Keirstead. Renewables, nuclear, or fossil fuels? scenarios for great britain's power system considering costs, emissions and energy security. *Applied Energy*, 152:83–93, 8 2015. ISSN 03062619. doi:10.1016/j.apenergy.2015.04.102.
- [35] Seán Collins, Paul Deane, Brian Ó Gallachóir, Stefan Pfenninger, and Iain Staffell. Impacts of inter-annual wind and solar variations on the european power system. *Joule*, 2:2076–2090, 10 2018. ISSN 25424351. doi:10.1016/j.joule.2018.06.020.
- [36] Iain Staffell and Stefan Pfenninger. The increasing impact of weather on electricity supply and demand. *Energy*, 145:65–78, 2 2018. ISSN 03605442. doi:10.1016/j.energy.2017.12.051.
- [37] Michael T. Craig, Jan Wohland, Laurens P. Stoop, Alexander Kies, Bryn Pickering, Hannah C. Bloomfield, Jethro Browell, Matteo De Felice, Chris J. Dent, Adrien Deroubaix, Felix Frischmuth, Paula L.M. Gonzalez, Aleksander Grochowicz, Katharina Gruber, Philipp Härtel, Martin Kittel, Leander Kotzur, Inga Labuhn, Julie K. Lundquist, Noah Pflugradt, Karin van der Wiel, Marianne Zeyringer, and David J. Brayshaw. Overcoming the disconnect between energy system and climate modeling, 7 2022. ISSN 25424351.
- [38] I. Tobin, W. Greuell, S. Jerez, F. Ludwig, R. Vautard, M. T.H. Van Vliet, and F. M. Breón. Vulnerabilities and resilience of european power generation to 1.5 °c, 2 °c and 3 °c warmingimpact of climate change on backup energy and storage needs in wind-dominated power systems in europe. *Environmental Research Letters*, 13, 4 2018. ISSN 17489326. doi:10.1088/1748-9326/aab211.
- [39] Juliane Weber, Jan Wohland, Mark Reyers, Julia Moemken, Charlotte Hoppe, Joaquim G. Pinto, and Dirk Witthaut. Impact of climate change on backup energy and storage needs in wind-dominated power systems in europe. *PLoS ONE*, 13, 8 2018. ISSN 19326203. doi:10.1371/journal.pone.0201457.
- [40] Jacek Kapica, Jakub Jurasz, Fausto A. Canales, Hannah Bloomfield, Mohammed Guezgouz, Matteo De Felice, and Kobus Zbigniew. The potential impact of climate change on european renewable energy droughts. *Renewable and Sustainable Energy Reviews*, 189, 1 2024. ISSN 18790690. doi:10.1016/j.rser.2023.114011.
- [41] Damien Raynaud, Benoit Hingray, Baptiste François, and Jean Dominique Creutin. Energy droughts from variable renewable energy sources in european climates. *Renewable Energy*, 125:578–589, 2018.
- [42] A. T.D. Perera, Vahid M. Nik, Deliang Chen, Jean Louis Scartezzini, and Tianzhen Hong. Quantifying the impacts of climate change and extreme climate events on energy systems. *Nature Energy*, 5:150–159, 2 2020. ISSN 20587546. doi:10.1038/s41560-020-0558-0.
- [43] Jeffrey A. Bennett, Claire N. Trevisan, Joseph F. DeCarolis, Cecilio Ortiz-García, Marla Pérez-Lugo, Bevin T. Etienne, and Andres F. Clarens. Extending energy system modelling to include extreme weather risks and application to hurricane events in puerto rico. *Nature Energy*, 6: 240–249, 3 2021. ISSN 20587546. doi:10.1038/s41560-020-00758-6.
- [44] S.I. Seneviratne, X. Zhang, M.Adnan, W. Badi, C. Dereczynski, A.D. Luca, S. Ghosh, I. Iskandar, J. Kossin, and S. Lewis. Climate change 2021: The physical science basis. contribution of working group i to the sixth assessment report of the intergovernmental panel on climate change. weather and climate extreme events in a changing climate, 2021.
- [45] ENTSO-E. European resource adequacy assessments, 2023. URL https://www.entsoe.eu/ outlooks/eraa/2023/report/ERAA_2023_Executive_Report.pdf.
- [46] Copernicus Climate Data Store. The copernicus climate data store, 2023. URL https://cds. climate.copernicus.eu/#!/home.

- [47] Iain Staffell and Stefan Pfenninger. Using bias-corrected reanalysis to simulate current and future wind power output. *Energy*, 114:1224–1239, 2016.
- [48] Intergovernmental Panel on Climate Change | Data Distribution Centre. What is a gmc? URL https://www.ipcc-data.org/guidelines/pages/gcm_guide.html.
- [49] Lee Hannah. Climate change biology. Academic Press, 2021.
- [50] H. C. Bloomfield, D. J. Brayshaw, A. Troccoli, C. M. Goodess, M. De Felice, L. Dubus, P. E. Bett, and Y. M. Saint-Drenan. Quantifying the sensitivity of european power systems to energy scenarios and climate change projections. *Renewable Energy*, 164:1062–1075, 2 2021. ISSN 18790682. doi:10.1016/j.renene.2020.09.125.
- [51] Christian M Grams, Remo Beerli, Stefan Pfenninger, Iain Staffell, and Heini Wernli. Balancing europe's wind-power output through spatial deployment informed by weather regimes. *Nature climate change*, 7(8):557–562, 2017.
- [52] Martin Wild, Doris Folini, Florian Henschel, Natalie Fischer, and Björn Müller. Projections of longterm changes in solar radiation based on cmip5 climate models and their influence on energy yields of photovoltaic systems. *Solar Energy*, 116:12–24, 2015.
- [53] Oliver Schmidt and Iain Staffell. *Monetizing energy storage: a toolkit to assess future cost and value*. Oxford University Press, 2024.
- [54] Paul Albertus, Joseph S. Manser, and Scott Litzelman. Long-duration electricity storage applications, economics, and technologies, January 2020. ISSN 25424351.
- [55] Paul Denholm. *The challenge of defining long-duration energy storage*. National Renewable Energy Laboratory, 2021.
- [56] Scott-Litzelman Max, Tuttman. Why long-duration energy storage matters, 2020. URL https: //arpa-e.energy.gov/news-and-media/blog-posts/why-long-duration-energy-storage-matters.
- [57] Jacqueline A. Dowling, Katherine Z. Rinaldi, Tyler H. Ruggles, Steven J. Davis, Mengyao Yuan, Fan Tong, Nathan S. Lewis, and Ken Caldeira. Role of long-duration energy storage in variable renewable electricity systems. *Joule*, 4:1907–1928, 9 2020. ISSN 25424351. doi:10.1016/j.joule.2020.07.007.
- [58] Nestor A. Sepulveda, Jesse D. Jenkins, Aurora Edington, Dharik S. Mallapragada, and Richard K. Lester. The design space for long-duration energy storage in decarbonized power systems. *Nature Energy*, 6:506–516, 5 2021. ISSN 20587546. doi:10.1038/s41560-021-00796-8.
- [59] AFRY. Resource adequacy in the 2030s, 2022. URL https://www.nationalgrideso.com/document/ 318151/download.
- [60] Bruno Cárdenas, Lawrie Swinfen-Styles, James Rouse, and Seamus D Garvey. Short-, medium-, and long-duration energy storage in a 100% renewable electricity grid: A uk case study. *Energies*, 14(24):8524, 2021.
- [61] Herib Blanco and André Faaij. A review at the role of storage in energy systems with a focus on power to gas and long-term storage, 2018. ISSN 18790690.
- [62] Felix Cebulla, Jannik Haas, Josh Eichman, Wolfgang Nowak, and Pierluigi Mancarella. How much electrical energy storage do we need? a synthesis for the us, europe, and germany. *Journal* of Cleaner Production, 181:449–459, 2018.
- [63] Jürgen Kepplinger, Fritz Crotogino, Sabine Donadei, and Manfred Wohlers. Present trends in compressed air energy and hydrogen storage in germany. In Solution Mining Research Institute SMRI Fall 2011 Conference, York, United Kingdom, 2011.

- [64] Sebastian Schiebahn, Thomas Grube, Martin Robinius, Vanessa Tietze, Bhunesh Kumar, and Detlef Stolten. Power to gas: Technological overview, systems analysis and economic assessment for a case study in germany. *International journal of hydrogen energy*, 40(12):4285–4294, 2015.
- [65] Markus Lehner, Robert Tichler, Horst Steinmüller, Markus Koppe, et al. *Power-to-gas: technology* and business models. Springer, 2014.
- [66] Christina Wulf, Jochen Linßen, and Petra Zapp. Review of power-to-gas projects in europe. *Energy Procedia*, 155:367–378, 2018.
- [67] Turgut M Gür. Review of electrical energy storage technologies, materials and systems: challenges and prospects for large-scale grid storage. *Energy & Environmental Science*, 11(10): 2696–2767, 2018.
- [68] Ronald N Allan et al. Reliability evaluation of power systems. Springer Science & Business Media, 2013.
- [69] Stefan Pfenninger and Bryn Pickering. Calliope: a multi-scale energy systems modelling framework. Journal of Open Source Software, 3(29):825, 2018.
- [70] Stefan Pfenninger and James Keirstead. Renewables, nuclear, or fossil fuels? scenarios for great britain's power system considering costs, emissions and energy security. *Applied Energy*, 152:83–93, 2015.
- [71] Stefan Pfenninger and James Keirstead. Comparing concentrating solar and nuclear power as baseload providers using the example of south africa. *Energy*, 87:303–314, 2015.
- [72] Tröndle Tim, Lilliestam Johan, Marelli Stefano, and Pfenninger Stefan. Trade-offs between geographic scale, cost, and infrastructure requirements for fully renewable electricity in europe. joule 2020; 4 (9): 1929–48.
- [73] Taco Niet. Storage end effects: An evaluation of common storage modelling assumptions. Journal of Energy Storage, 27:101050, 2020.
- [74] R. Billinton and Peng Wang. Teaching distribution system reliability evaluation using monte carlo simulation. IEEE Transactions on Power Systems, 14(2):397–403, 1999. doi:10.1109/59.761856.
- [75] Wenyuan Li et al. *Reliability assessment of electric power systems using Monte Carlo methods*. Springer Science & Business Media, 2013.
- [76] Nils Ohlendorf and Wolf-Peter Schill. Frequency and duration of low-wind-power events in germany. *Environmental Research Letters*, 15(8):084045, 2020.
- [77] Panit Potisomporn, Thomas AA Adcock, and Christopher R Vogel. Extreme value analysis of wind droughts in great britain. *Renewable Energy*, 221:119847, 2024.
- [78] European Network Transmission System Operator. Visualisation platform, 2022. URL https: //2022.entsos-tyndp-scenarios.eu/visualisation-platform/.
- [79] Global Wind Atlas, 2024. URL https://globalwindatlas.info/en.
- [80] Pablo Ruiz, Wouter Nijs, Dalius Tarvydas, Alessandra Sgobbi, Andreas Zucker, Roberto Pilli, Ragnar Jonsson, Andrea Camia, Christine Thiel, Carsten Hoyer-Klick, et al. Enspreso-an open, eu-28 wide, transparent and coherent database of wind, solar and biomass energy potentials. *Energy Strategy Reviews*, 26:100379, 2019.
- [81] MD Felice, G Peronato, and K Kavvadias. energy-modelling-toolkit/hydro-power-database: Jrc hydro-power database-release 10. zenodo, 2021.
- [82] Laurens P. Stoop and Jing Hu. Hydropower dataset of hourly inflow values for european bidding zones for acdc-esm, 2023. URL https://zenodo.org/records/7766457.

- [83] Oliver Schmidt, Sylvain Melchior, Adam Hawkes, and Iain Staffell. Projecting the future levelized cost of electricity storage technologies. *Joule*, 3(1):81–100, 2019.
- [84] Piergiorgio Alotto, Massimo Guarnieri, and Federico Moro. Redox flow batteries for the storage of renewable energy: A review. *Renewable and sustainable energy reviews*, 29:325–335, 2014.
- [85] Danish Energy Agency. Technology catalogues, 2024. URL https://ens.dk/en/our-services/ technology-catalogues.
- [86] LACAL ARANTEGUI Roberto, JAEGER-WALDAU Arnulf, VELLEI Marika, SIGFUSSON Bergur, MAGAGNA Davide, JAKUBCIONIS Mindaugas, PEREZ FORTES Maria Del Mar, LAZAROU Stavros, GIUNTOLI Jacopo, WEIDNER RONNEFELD Eveline, et al. Etri 2014-energy technology reference indicator projections for 2010-2050. 2014.
- [87] Tobias Boßmann and Iain Staffell. The shape of future electricity demand: Exploring load curves in 2050s germany and britain. *Energy*, 90:1317–1333, 2015.
- [88] European Network System Operators. Download, 2022. URL https://2022. entsos-tyndp-scenarios.eu/download/.
- [89] Frank A Felder and Marie Petitet. An emerging framework for the probabilistic cost-benefit analysis of the reliability, resiliency, and adaptability of electric power systems. *Resiliency, and Adaptability of Electric Power Systems (June 15, 2024)*, 2024.
- [90] Gregory Peter et al. The value of lost load (voll) in european electricity markets: uses, methodologies, future directions. In 2019 16th International Conference on the European Energy Market (EEM), pages 1–6. IEEE, 2019.
- [91] Will Gorman. The quest to quantify the value of lost load: A critical review of the economics of power outages. *The Electricity Journal*, 35(8):107187, 2022.
- [92] Electricity System Operator ESO. Resource adequacy in the 2030s spotlight: Exploring approaches and metrics to assess resource adequacy in a fully decarbonised power system, 2024. URL https://www.nationalgrideso.com/document/318151/download.
- [93] Solomon Gebre, Knut Alfredsen, Leif Lia, Morten Stickler, and Einar Tesaker. Review of ice effects on hydropower systems. *Journal of Cold Regions Engineering*, 27(4):196–222, 2013.
- [94] Energy Systems Integration Group. A report on the redefining resource adequacy task force, 2021. URL https://www.esig.energy/wp-content/uploads/2021/08/ ESIG-Redefining-Resource-Adequacy-2021.pdf.
- [95] Sebastian Gonzato, Kenneth Bruninx, and Erik Delarue. On the economic justification for the loss of load expectation in the presence of energy limited resources, 2018.
- [96] Paul Denholm, Wesley Cole, A Will Frazier, Kara Podkaminer, and Nate Blair. The four phases of storage deployment: A framework for the expanding role of storage in the us power system. *Golden, CO*, 2021.
- [97] Regelleistung Online. Batteriespeicher dominieren den prl-markt, 2020. URL https://www. regelleistung-online.de/batteriespeicher-dominieren-den-prl-markt/.
- [98] Arild Helseth, Albert CG Melo, Quentin M Ploussard, Birger Mo, Maria EP Maceira, Audun Botterud, and Nathalie Voisin. Hydropower scheduling toolchains: Comparing experiences in brazil, norway, and usa and implications for synergistic research. *Journal of Water Resources Planning* and Management, 149(7):04023030, 2023.
- [99] Robert Zajdler. The european union resource adequacy assessment as an instrument to support the development of renewable energy sources and the achievement of decarbonisation targets. *Przegląd Ustawodawstwa Gospodarczego*, (2):2–9, 2022.
- [100] David Parra and Romain Mauger. A new dawn for energy storage: An interdisciplinary legal and techno-economic analysis of the new eu legal framework. *Energy Policy*, 171:113262, 2022.

[101] Eric Cutter, Ben Haley, Jeremy Hargreaves, and Jim Williams. Utility scale energy storage and the need for flexible capacity metrics. *Applied energy*, 124:274–282, 2014.